

UPB-CSUD Decision no. 975/ 08.12.2022

Flavia-Petruța-Georgiana G. ARTIMON (STOCHIOIU)

PhD THESIS *Summary*

Contributions to structure and attributes development of laser interferometry systems

DOCTORATE COMMISSION

President	Prof.univ.dr.ing. Liviu- Daniel GHICULESCU		
PhD Supervisor	Prof.univ.dr.ing. Marian GHEORGHE	University POLITEHNICA of Bucharest	
Referee	Prof.univ.dr.ing. Nicolae IONESCU		
Referee	Prof.univ.dr.ing. Viorel PĂUNOIU	University "Dunărea de Jos" of Galați	
Referee	Dr.ing. Mihai MĂRGĂRITESCU	National Institute of Research and Development in Mechatronics and Measurement Technique (INCDMTM)	

Content

Summary: S; Thesis: T	S	Т
Foreword	3	3
Introduction	4	4
List of abbreviations	5	5
Part I. The actual state of the art on measuring processes and systems		
Chapter 1. Measuring processes and production systems	7	7
1.1. Metrology, quality control and standards	7	7
1.1.1. Measuring process and characteristics	7	10
1.1.2. Measurement errors	8	11
1.2. Production and measuring process inspection	8	13
1.3. Calibration of production and measuring systems	9	14
1.4. Factors affecting measuring and production processes	9	16
1.5. Influence of the temperature as an environmental factor	9	17
Chapter 2. Measuring equipment and devices	11	21
2.1. Introduction	11	21
2.2. Classification	11	21
2.2.1. Classification based on the output results	11	21
2.2.2. Classification based on the input data	11	22
2.2.3. Classification based on precision	11	23
2.3. Interferometry and interferometers	11	24
2.3.1. Basic principle	11	24
2.3.2. Types of interferometers based on the division method	11	27
2.3.3. Common types of interferometers	13	39
2.3.4. Uncertainty sources in laser measurements	14	31
2.3.5. Laser interferometers for machine tools control and calibration	14	33
2.3.6. Grippers and mounting kits available on the market for interferometers	15	39
2.3.7. Additive manufacturing of grippers and mounting parts	15	40
Chapter 3. Conclusions regarding the state of the art on measuring processes and systems	18	43

PhD Thesis *Summary*

Part II. Contributions to defining and constructive development of laser interferometry systems structure

<i>Chapter 4.</i> The directions, main objective and methodology of research and development on laser interferometry systems structure	22	47
4.1. Directions of research and development	22	47
4.2. The main objective of research and development activity	22	47
4.3. Methodology of research and development	22	47
Chapter 5. General structure and characteristics of laser interferometry systems	25	51
Chapter 6. Development of an innovative laser interferometry system for linear position	27	59
6.1. Innovative structure of a laser interferometry system for linear position	27	59
6.2. Conceptualization and detailed design of an innovative splitter positioning device	27	61
6.2.1. Initial data and conceptualization of the splitter positioning device	27	61
6.2.2. Detailed design of the splitter positioning device	27	62
6.3. Manufacturing on the splitter positioning device	27	72
6.3.1. Pre-processing for additive manufacturing	27	72
6.3.2. Additive manufacturing on the splitter positioning device	27	74
6.3.3. Post-processing after additive manufacturing	27	76
<i>Chapter</i> 7. Validation of the new splitter positioning device within laser interferometry systems for linear position	31	79
7.1. Experimental protocol and environmental conditions	31	79
7.1.1. Experimental protocol	31	81
7.1.2. Study on environmental conditions	31	84
7.2. Consecutive tests in laboratory conditions	32	85
7.3. Parallel tests in laboratory conditions	33	88
7.4. Parallel tests in industrial conditions	34	91
<i>Chapter</i> 8. Final conclusions and main contributions regarding the structure and attributes development of laser interferometry systems	37	97
References	41	101

Foreword

The research and development on the laser interferometry systems represent the *motivation* and *direction* of the doctoral studies, finalised by this doctoral thesis.

The doctoral activity consisted of preparation, presentation and defence of exams and scientific reports, profound study, development of defining analytical descriptors for general structure and characteristics of laser interferometry systems, proposal, design, manufacturing and validation of an innovative splitter positioning device, as part of a laser interferometry system for linear position, creating and publishing of scientific papers, as well as the elaboration of the present doctoral thesis concerning the contributions to structure and attributes development of laser interferometry systems

First and foremost, I would like to thank and praise God for making this thesis possible, and to everyone who carried me into their prayers.

I address my profound gratitude to Prof.univ.dr.ing. Marian GHEORGHE, for his scientific guidance and coordination during the entire activity of completing this doctoral program, for the support and full trust granted during this period.

I address my sincere thanks to Conf.univ.dr.ing.mat. Ovidiu BLĂJINĂ, Şef lucr.univ.dr.ing. Daniel MANOLACHE and CS1 dr.ing. Gheorghe POPAN, for their reviews and recommendations as being a part of the Advisory Committee for the doctoral scientific reports and doctoral thesis preliminary defence.

I express my appreciation and respectful thanks to Prof.univ.dr.ing. Liviu-Daniel GHICULESCU, from University POLITEHNICA of Bucharest, Prof.univ.dr.ing. Viorel PĂUNOIU – University "Dunărea de Jos" of Galați, CS1 dr.ing. Mihai MĂRGĂRITESCU - National Institute of Research and Development in Mechatronics and Measurement Technique (INCDMTM), Prof.univ.dr.ing. Nicolae IONESCU - University POLITEHNICA of Bucharest, for their comments and recommendations, as president and members of the official Doctorate Commission for scientific assessment and examination during public presentation of the PhD Thesis.

Special thanks to Ing. Nasim GANNAM and Ing. Cătălin-Ștefan NEGREA for their support to prepare and operate machining numerical centres tested through the developed laser interferometry system within my doctoral research.

I will be forever grateful to my loving brother whose words of advice never cease to amaze me with their accuracy.

This thesis is dedicated to my father and mother who raised me beautifully, always care for me and love me unconditionally. It is also dedicated to my husband, who loved, encouraged and supported me during this difficult period of time.

Flavia-Petruta-Georgiana G. ARTIMON (STOCHIOIU)

Introduction

Production and measuring processes are co-dependent. Their evolution can be seen worldwide through a continuous improvement of production quality, capacity and efficiency.

The measuring processes and production systems require high precision in order to meet the product specifications. This leads to the requirement that the equipment for quality control to be able of micrometre/ nanometre scale measurements.

Since its invention in the 19th century, the interferometry method has seen a continuous development and has known a large variation of applications. Devices based on this method are some of the most precise used in measurements, in all fields of technology. For evaluating and measuring processes and production systems, it is a widely accepted and proven technology, with its nanometric precision capabilities. A restricting factor, for current laser interferometry systems, is the requirement that certain optical elements to be mounted inside the calibrated machine workspace, thus limiting the measuring range. This doctoral thesis is aimed at analysing and proposing a solution to this issue, by introducing a system structure development.

* * *

The first part of the thesis presents the state of the art on measuring processes and systems. With the aim of being aware of the various factors that may intervene into the quality results of the production and measuring systems, the metrology, quality control and standards concepts are elaborated. Further, classifications of the existing measuring devices are made and some devices are detailed. The interferometric concept, procedure and components are detailed for further studies on their capabilities.

In accordance with the data and conclusions on the state of the art research, the interferometry drew the attention, so that the main objective of the research activity was determined as: structure and attributes development of laser interferometry systems.

A development of defining analytical descriptors for general structure and characteristics of laser interferometry systems is proposed, at high level of generality, open to particularizations, in context, as well to other description levels.

According to the interferometry limitations previously studied, a new splitter positioning device for laser interferometry systems is conceptualized with respect to the functional and constructive characteristics required. Thus, several iterations of the product are considered, in order to obtain an optimal design. The final iteration has lattice structure integrated, with all characteristics allowing its usage with various interferometers. In order to manufacture the splitter positioning device, the design is optimized for direct metal laser sintering out of CoCrSP2 superalloy powder.

For the experimental tests, two similar laser interferometry systems are used, in consecutive tests, and in parallel tests, respectively, on different CNC machining centres, one in a laboratory environment, and other two in industrial conditions, in order to validate the capabilities of the considered systems and, implicitly, of the new developed splitter positioning device.

As potential perspective, a complex research should continue on interacting processes, analytical modelling and operational characteristics associated to laser interferometry systems, in order to develop their applications in active industrial environments.

List of abbreviations

No.	Abbrev.	Significance		
01	ADC	Analog to Digital Converter		
02	ASRO	The Romanian Association of Standardization		
03	CAD	Computer Aided Design		
04	CEN	European Committee for Standardization		
05	CENELEC	European Committee for Electrotechnical Standardization		
06	СММ	Coordinate Measuring Machines		
07	CNC	Computerized Numerical Control		
08	DAC	Digital to Analog Converter		
09	DMLS	Direct Metal Laser Sintering		
10	ECG	Electrocardiogram		
11	EDM	Electrical Discharge Machining		
12	ETSI	European Telecommunications Standards Institute		
13	FEA	Finite Element Analysis		
14	FEM	Finite Element Method		
15	HRC	Rockwell Hardness		
16	IEC	International Electrotechnical Commission		
17	ISO	International Organization for Standardization		
18	LCD	Liquid Crystal Display		
19	LED	Light Emitting Diode		
20	LIS	Laser Interferometric System		
21	PA (2200)	Polyamide (2200)		

22	PSC	Process Software and Computer(s)	
23	QC	Quality Control	
24	SLS	Selective Laser Sintering	
25	SPD	Splitter Positioning Device	
26	TQM	Total Quality Management	
27	VMC	Vertical Machine Centre	
28	WCN	Wavelength Compensation Number	
30	3D	Three Dimensional	

Part I. The actual state of the art on measuring processes and systems

Chapter 1. Measuring processes and production systems

1.1. Metrology, quality control and standards

Generally, production and measuring processes for each individual product must be punctiliously prepared in order to achieve the required technical and economical product criteria.

When it comes to a production process, a general causeand-effect diagram of this concept was determined (Fig. 1.1) [I01].

The process that ensures that the production and measuring systems are working in the stipulated parameters is quality control.

Quality control cannot be accomplished without measurements. This concept also comprises establishing and forestalling the cause for which a product could be identified as non-compliant. Inspection and calibration are parts of quality control that encompass the scope of keeping the process up to the specific quality characteristics determined initially, through measurements [W01].

Inspection is an important aspect of quality control. Moreover, quality assurance includes all aspects of quality control and inspection, as it contains quality planning, quality inspection and quality supervision. Shortly, quality assurance is the sum of measures taken to ensure quality [M01].

Depending on their purpose, there are different types of standards oriented towards referencing, calibration, inspection or working. A brief hierarchical classification of standards is depicted in Fig. 1.3.

1.1.1. Measuring process and characteristics

Choosing the measuring principle depends on many factors, out of which the most important could be considered the measurand characteristics, required precision, resolution, stability, environment, time, costs etc. Taking this into consideration, throughout time, numerous types of measurement procedures have been explored.

Choosing the perfect measuring device cannot be possible due to the fact that there are no ideal instruments for this type of activity and usually, the device that might offer the best response is noticeably expensive [R01]. This is one of the reasons why choosing the measuring device needs to take into account the scope of the measurement. Whether it is for study on a site, in a laboratory or in a factory, even though the measurand might be the same, the proper device might differ.



Fig. 1.1. Cause-and-effect diagram [I01]



Fig. 1.3. Hierarchical classification of standards [R02]

1.1.2. Measurement errors

A measurement error can be summarized as the difference between the measured quantity and the absolute value of that quantity. As the absolute value cannot be truly known, the real measurement error is the sum of the device measurement errors, methodological errors, human errors and statistics [G02].

While the classification can be procedure-based, errors can also be distinguished based on the factors that cause their characteristics: systematic and random. Systematic errors comprise the device, environmental and observational errors, and influence the accuracy of the measurement, while random errors are frequently caused by an unexpected variation of the work environment and influence the precision of the measurement results. [R01]

Mathematically, the relationship between accuracy/inaccuracy, repeatability and systematic errors can be described as [G02]:

$$Accuracy = \sqrt{(repeatability)^2 + (inaccuracy)^2 + (systematic \ errors)^2}$$
(1.1)

A Gaussian description of random and systematic errors is depicted in Fig. 1.7. As it may be noticed, the actual result of a measurement may be written as [G03]:

$$Result = true \ value + systematic \ error + random \ error \tag{1.2}$$



Fig. 1.7. Random and systematic errors description [adapted from I06]

Unlike random errors, systematic errors are easy to notice and evaluate, cannot be eliminated by increasing the number of measurements but can be reduced by device calibration. Moreover, random errors are characterized by variance, standard deviation and mean, and can be both positive and negative [R02].

1.2. Production and measuring process inspection

The sum of tests and verifications performed on measuring and production equipment represents an inspection procedure [B01]. When performing an inspection on an equipment, guidelines may be provided by the manufacturer but, usually, this procedure is detailed by approved guidelines [P01]. Companies tend to develop specific technical norms for production and measuring process control in order to meet the client's needs, the equipment capacities, the staff qualifications and product specifications.

Choosing the appropriate method for inspection depends on the desired accuracy. The two are linked through the economic aspect as, an increase in accuracy usually is accompanied in an increase in costs, which can have an exponential evolution

In order to perform a production inspection, the instrument chosen to be used has to be calibrated with a device having ten times the accuracy [R02, C03].

1.3. Calibration of production and measuring systems

The calibration process combines a suite of operations in special environmental conditions in order to establish the relationship between a measuring system and the measurand and to eliminate the systematic errors [B02].

This procedure is applied mostly to devices, instruments and equipment of production process and metrology. Calibration is performed under specific conditions determined by standards and it refers to direct comparison of output values and the previously known dimensions or performance of the measurand.

An authorized person may perform this process and it leads to a certificate issue. If the output values exceed the tolerance values, a recalibration procedure should be operated.

One of the most important attributes of the measuring instruments is traceability. То achieve traceability, the calibration of the measuring devices and equipment by following international standards is crucial [F01].

To facilitate this traceability, international standards, considered primary, are adapted into national standards (by ASRO, for example), widely known as secondary, and to working standards (Fig. 1.11).



1.5. Influence of the temperature as an environmental factor

Temperature instabilities may produce extreme differences when micrometric accuracy is needed, therefore certain solutions [R04] were developed to overcome these problems. Generally, approaches focus on the sources of temperature fluctuation.

In the case of a machine-tool there are five origins of this problem: room environment, coolants, people, machine-tool and cutting process [B03]. Each source transfers heat through one of the three possible ways: conduction, convection or radiation. This heat transfer might be uniform, such as in the case of room environment or coolants sources, or non-uniform from other causes.

Temperature variations may affect dimensions of the operating device (master and machine frame) and the part/ measurand. This disturbance may produce either contractions or expansions to the material that is formulated as a multiplication of the initial length, the expansion coefficient and the temperature [V02].

The expansion coefficient indicates the deformation type by use of a positive or negative sign before the value number. Contraction is depicted in a negative form, while the expansion is positive.

In order to illustrate the temperature influence on the linear coefficient, several results from the literature [T01] are gathered in Fig. 1.14. It is noticeable that most of the listed metals behave in a similar fashion, with an increase in the expansion coefficient, with the exception of Invar, also known as FeNi36, which has a particular evolution with respect to temperature.

This iron alloy has a very low thermal expansion, at temperatures up to over 100° C. This is due to the magnetic and electronic contributions, which dominate the behaviour at low temperatures. Therefore, this material is frequently used in the construction of measuring equipment and others that require dimensional stability.



Fig. 1.14. Thermal linear coefficient for various metallic alloys

Production and measuring equipment manufacturers are continuously improving their products and the decrease of temperature influence is one of the focused improvements.

In order to avoid thermal drift during a measuring procedure, standards such as [I10] recommend soaking both the measurand and the measuring instrument in a 20° C stable environment, hence the construction of each device contains different types of materials, therefore different expansions or contractions may affect the process.

Chapter 2. Measuring equipment and devices

2.1. Introduction

A measuring instrument is a set of technical means having the scope of establishing a quantity or a variable, considering national and/or international standards. Generally, this type of instruments is equipped with sensors and/or transducers, signal conditioning division and a signal processing stage, as presented in Fig. 2.1 [W03].



Fig. 2.1. General structure of a measuring instrument [W03]

Every device compels periodic calibration to maintain quality. When referring to the production machines calibration or even measuring equipment calibration, the most precise devices, such as interferometers, autocollimators, electronic levels, reference encoders, etc. might be required.

2.3. Interferometry and interferometers

Throughout time, a multitude of measurement techniques and instruments were developed, and their accuracy has improved constantly.

At this moment, one of the most accurate measurement devices is the interferometer [B06]. Albert Michelson was the first to implement interferometry in measurement in the late 19th Century and the working principle of all the interferometers developed until today was slightly changed.

This measurement technique is versatile, commonly being used in high quality measurements.

Interferometry has its roots into the wave theory of light which, more than 300 years ago, explained that the light behaves as waves, similar to sound [R05]. Starting from the indispensable theories of light and wave, the fundamentals of the interferometry method are derived. Light waves are composed of perpendicular electric and magnetic waveforms (Fig. 2.8).

Each (harmonic) wave may be described by five parameters: irradiance, phase angle, propagation direction, wavelength and direction of polarization [D02].



Fig. 2.8. Electromagnetic wave [B07]

Those parameters change when the beam meets the measuring

object. Interferometry captures those changes and studies them to acquire information about the measurand.

(a) Irradiance

Irradiance, also known as flux density of a wave, is the total energy that flows per second across a unit area perpendicular to the propagation direction. In interferometry, irradiance is also entitled as the beam intensity, being considered the beam's capacity to concentrate the energy into a specific area within a certain time. This parameter is very important especially in domains where it is used for cutting, welding or heat treatments (surgeries, metal products manufactories, etc.). Irradiance is measured in W/m² and it is proportional to the square of the wave's amplitude [R06, P03].

	PhD Thesis	Part I. The actual state of the art on measuring	Flavia-Petruța-Georgiana G. ARTIMON
UPB	*Summary*	processes and systems	(STOCHIOIU)

(b) Phase angle

The phase angle is the angular displacement of a sinusoidal wave from a given reference. It can only be measured by means of the superimposition phenomenon. Therefore, if two waves are traveling in the same direction with the same amplitude, frequency, and velocity, but one has a phase angle difference of ϕ , their equations [S04] are:

$$Y_1 = A\sin(kx - wt - \phi)$$
(2.1)

$$Y_2 = A\sin(kx - wt) \tag{2.2}$$

where: A is amplitude; $k = 2\pi/\lambda$; ω - the angular frequency ($\omega = 2\pi\vartheta$); x - the direction; t - time. If those two waves are superimposed, their waveform is given by:

$$Y = 2A\cos\left(\frac{\Phi}{2}\right)\sin\left(kx - wt - \frac{\Phi}{2}\right)$$
(2.3)

As mentioned earlier, irradiance is proportional to the square of the wave's amplitude, therefore:

$$I = 4I_0 \cos\left(\frac{\phi}{2}\right) \tag{2.4}$$

where, I_0 is the irradiance of one of the two waves.

When two identical waves superpose in phase, the field strength doubles while when the two waves oscillate in opposite phase (superimpose), they equal out each other (Fig. 2.9) due to the fact that their field strength is calculated by summing the vectors of each field strength. In consequence, in the first case, the irradiance multiplies four times, while in the second case is null.



Fig. 2.9. Wave interference [I11]

(c) Propagation direction

An obvious definition is that the direction of the propagation is the travelling axis of the wave.

However, when two waves differ from each other only by their directions of propagation they superimpose creating an interference pattern that allows various measurements. The time interval between two peaks of the fringes, D, may be determined knowing the angle, θ , between the two waves, and their wavelength, λ , as:

$$D = \frac{\lambda}{2\sin\theta} \tag{2.5}$$

(d) Wavelength

Wavelength is the distance between a start point and an end point of a complete oscillation cycle. Since the sine waves are symmetric, the wavelength might also be defined as the distance between two similar points in the wave (Fig. 2.8). Previously, it has been established the importance of accuracy and repeatability, especially in precise measurements. Wavelength needs a frequency stabilization in order to provide those two characteristics, therefore all the interferometric laser sources have a stabilization mechanism [S05]. The output of an interferometric measurement are fringes. When a displacement occurs in the system, the fringes are disturbed. Each fringe represents a division of a wavelength. In order to measure those displacements, wavelength measurements, based on the superposition principle, can be approached. Being considered two waves, E_1 and E_2 , with different frequencies that are superimposed. The two wavelengths, λ and $\lambda + \Delta\lambda$, lead to a resulted wave that has an irradiance modulated in time. The time interval between two peaks of the irradiance are entitled beat periods, T_b , and:

$$T_b = \frac{\lambda^2}{\frac{\omega}{k} \Delta \lambda}$$
(2.6)

where: ω is the angular frequency, and k - the wave number.

This equation helps to identify the wavelength shift, considering an established absolute wavelength. If the measurement period is smaller than a beat period, then the beat period cannot be determined.

(e) Direction of polarization

The polarization direction is given by the transverse direction of the electric field. Generally, since the direction of propagation is considered to be along the Z axis, the light polarization is taking place in the X-Y domain (Fig. 2.14, a).

In the basic stages of interferometry, the polarization direction is linear (Fig. 2.14. b) but, usually, it acquires a more complex, elliptic or circular form. These complex forms have a 90° phase lag and their polarization direction divides them in two categories: right-hand/counter-clockwise polarized or left-hand/clockwise polarized (Fig. 2.14. c).



Fig. 2.14. Beam polarization direction [E02]

Light can be defined as an electromagnetic wave traveling through space. If a linearly polarized wave, is propagating along a z direction, its equation may be represented as follows:

$$E = \alpha \cos \left[2\pi \left(\vartheta t - z/\lambda \right) \right] \tag{2.7}$$

where: α is the amplitude of the light wave; ϑ - the frequency; λ - the wavelength.

2.3.3. Common types of interferometers

Interferometry has been of particular interest endorsing various applications such as: identifying a new method of defining the meter [R07], plasma diagnostics [V03], precision industrial measurement of lengths and movement [J01, S06, S07], linear deviations and strain measurements [A04, T03, X01], gravitational antennas [K05], vibration analysis [Z01, C09], illegal mining and landslides monitoring [M05, S08], blood speed



Fig. 2.18. Michelson interferometer set-up

measurement [I12, X02], diseased cells detection [N01] etc.

UPB PhD Thesis Pa *Summary*

Since Michelson demonstrated the first concept of interferometer by means of mirrors, a white light source, a compensation glass and a beam splitter (half-silvered mirror), countless interferometry principles, procedures and set-ups were developed. The linear set-up of a Michelson interferometer is depicted in Fig. 2.18. The laser beam is split and directed towards two mirrors that return the beams to the splitter and further they are reunited to the screen.

The output of this light travel is entitled interference fringes, meaning light sinusoidal intensity variation. In the Michelson's interferometer, if the distance between the beam splitter and one of the mirrors changes its value, Haidinger fringes (Fig. 2.19, (a)) appear on the screen, while if one of the mirrors slightly changes it is angle to the beam splitter, the result to be interpreted will be consisting of Fizeau fringes (Fig. 2.19, (b)) [M06].



Fig. 2.19. Interference fringes:a) Haidinger fringes [S09];b) Fizeau fringes [K06]

2.3.4. Uncertainty sources in laser measurements

Laser interferometry measurements uncertainty is mostly determined by the environmental errors (wavelength and thermal errors) and geometric errors (dead path, cosine and Abbe errors), as noted in the further equations [L05]:

$$e = L x (1 + \alpha \theta) - L_{LI} + e_{cos} + e_{DeadPath} + e_{Abbe}$$
(2.8)

$$u_{tot}^2 = \sum_i (u_i \ x \ c_i)^2 \tag{2.9}$$

where: u_i is standard uncertainty; c_i – sensitivity coefficient; $u_i x c_i$ – standard uncertainty contribution; L – measured probe displacement; e_{Abbe} – Abbe error; α – temperature expansion coefficient of the stage; θ – temperature deviation from 20°C; L_{LI} – laser interferometer measured distance; e_{cos} – cosine error; $e_{DeadPath}$ – dead path error; u_{tot} – total standard uncertainty.

2.3.5. Laser interferometers for machine tools control and calibration

A typical set-up for the calibration of a machine tool (Fig. 2.29) consists of a beam source, beam splitter(s), mirrors, a receptor, a support for the beam source (tripod), optic mounting kits and specific software. According to the measurement type, other supports, optic elements, environmental sensors and a compensator unit might be necessary. However, this set-up can be problematic because one of the optics is usually mounted on the main spindle. The intersection of the machine's movement limits axes the capabilities measuring by decreasing the measurable area.



Fig. 2.29. Common set-up of an interferometer for a linear measurement [B12]

Interferometers are widely used for calibration due to their high accuracy and reliability [L08] and are able to perform measurements of linear and angular positioning, straightness, flatness, squareness and parallelism. It is important to mention that, although interferometry is well appreciated in all of these applications, each of them has to be made with specific optical components, the whole set-up procedure and realignment of the laser beam being time consuming. Furthermore, as previously mentioned, the set-up of the optical elements might determine the measuring range to not cover the whole length of the evaluated axis.

2.3.6. Grippers and mounting kits available on the market for interferometers

The marketed interferometers are commonly accompanied by a mounting kit that provides the connection between the optics and the machine tools. The kit aims to allow an interchange ability for the optic parts with no beam realignment.

For instance, the Renishaw interferometer is equipped with mounting pillars, adaptors, base plates, clamp blocks and screws as presented in Fig. 2.38. The mounting pillars are designed to be screwed to a magnetic base while the clamp blocks ensure the connection between the magnetic base and the optic elements.

UPB

As manufacturers stated, the basic metallic parts (pillars/ columns and plates/ bases) are made out of magnetic or non-magnetic stainless steel [D03, R10].



Fig. 2.38. Optics mounting kit of Renishaw interferometer

2.3.7. Additive manufacturing of grippers and mounting parts

The additive manufacturing technology progress determined an increasing interest for lattice structures which are considered to be light-weight and offer control of flexibility. They began being used in different fields such as biomedical, aerospace and auto and continue being studied for various other industries. Some of the technical features were determined through the studies performed for cellular materials in previous decades, but the characteristics of lattice structures are still being researched due to their versatility.

In recent times, lattice structures began to be introduced in the construction of clamping and mounting systems and grippers, such as the systems presented in Fig. 2.40.



Fig. 2.40. Lattice structures within: a) fixing parts [E03]: b) robotic gripper [T09]

The main characteristic of lattice structures is that in the majority of cases, the only manufacturing method possible is 3D printing.

In order to be 3D printed, the parts containing lattice structures are digitally generated by means of different CAD software. Solidworks, Inventor and CATIA are virtual tools for building and designing from square one the lattice cells. It is then usually multiplied and subjected to a FEM analysis. Other software, such as Ansys, have started implementing modules which offer the possibility of generating pre-existing lattice structures (Fig. 2.41) in designated areas of the part/assembly.

Thus far, several characteristics have been determined for some widely known lattice structures such as cubic, diamond and octet [M09, D04, L11].

Since 2019, Ansys programme provides the possibility of lattice topological optimization, which enables anisotropic distribution of lattice construction thus being able to approach the models in

nature, found in bones, wings, leaves etc. (Fig. 2.42).

In nature, these lattice structures are the results of evolution, where different parts have adapted to the specific loads that they were to carry, while also respecting geometry constraints. The lattice unit cell dimension, its strut height and thickness, and the density distribution are therefore built according to the fixtures, loads and material characteristics.

In terms of materials used for fabricating metrological parts, including specific supports, Invar is the most recommended material, due to its thermal stability up to a temperature above 100° C (see Table 2.2), but stainless-steel material is frequently used for mounting and gripping elements in metrology [D03, R10, D05, A07]. The Elastic modulus, being considerably higher, resulting in more rigid structures, makes stainless steel more suitable for these types of application. А similar material to stainless steel is Inconel, which is known to be having better corrosion resistance and thermal stability; however, it is more expensive.

b) Cube lattice with Cube lattice with Regular cube lattice side diagonal supports center supports Cube lattice with side cross supports Octahedral lattice1 Octahedral lattice 2 Cube lattice with bottom Cube lattice with Double pyramid lattice center without vertical supports bottom center and face diagonals Double pyramid lattice with cross Diamond lattice Double pyramid Diamono lattice

Fig. 2.41. Lattice structures: b) Available lattice cells in ANSYS 2021 R1 programme



Fig. 2.42. Lattice structures found in nature: a) bone structure [N02]; b) insect wings [W06]; c) leaf ribs [W06]

UPB	PhD Thesis	Part I. The actual state of the art on measuring	Flavia-Petruța-Georgiana G. ARTIMON
UID	*Summary*	processes and systems	(STOCHIOIU)

Another plausible material for manufacturing is considered to be aluminium alloy, due to low costs and weight, and easy manufacture. However, in metrology, alloys consisting mainly of aluminium are usually avoided due to a generally high thermal expansion coefficient and low rigidity, as seen in Table 2.2. Rigidity or specific stiffness, K_s, is [A08]:

$$K_s = E/\rho \tag{2.16}$$

where: E is elastic modulus (tensile modulus or Young modulus), and ρ - density.

Material name	Thermal expansion coefficient [10 ⁻⁶ m/(m °C)]	Elastic modulus [GPa]	Density [g/cm ³]	Reference
FeNi36	1.2	137	8.1	[A10]
316L Stainless steel	16.3	193	8	[A11]
AlSi10Mg	20	70	2.67	[E04]
Inconel 718	12.8	205	8.22	[H04]
TiAl6V4	9	110	4.43	[E05]
CoCrSP2 superalloy	13.6	200	8.3	[E06]

Table 2.2. Characteristics of several materials for additive manufacturing

Notwithstanding, Al is commonly used in Ti based alloys for bone implants in human. With a similar rigidity to Stainless steel, but a better thermal stability and being biocompatible, the Ti alloys meet the requirements for bone replacement surgeries in almost any area of the body. CoCrSP2 is also biocompatible, but it frequently serves to dental implants.

Taking into consideration the rigidity of the materials listed in Table 2.2, it can be deduced that 316 L Stainless steel, Inconel 718, TiAl6V4 and CoCrSP2 superalloy are similar. However, the thermal expansion coefficient characteristics show a better behaviour for the last three of them.

Chapter 3. Conclusions regarding the state of the art on measuring processes and systems

From the analysis of the actual state of the art on measuring processes and systems, important conclusions can be drawn, as follows.

• Metrology is characterized by a variety of definitions, but it may be resumed as the study of measurements. It is a branch of science with its main parts: theoretical metrology, such as accuracy of measuring instruments, measurement theory, assurance for unity of measurements and fundamental concepts; particular metrology such as length, time, electric quantities, mechanical quantities, light, velocity, etc.; applied metrology, such as control and calibration of measuring instruments, certification of standard specimens, etc. (see § 1.1).

Various factors may interfere in a production and measuring system therefore, in order to provide compliant products, an entire process that evaluates and corrects eventual errors was developed and it is found in the literature as quality control. An important segment of quality control is referring to measurements which are usually carried out by means of norms and standards. Standards may be from a shop floor level to the national/international level and they usually focus on specific areas of inspection, calibration, work operations, etc. The standards assure traceability; therefore, each country adheres to a specific standardization organization. ASRO is a member of the European organizations CEN, CENELEC, ETSI and of the international organisations ISO and IEC (see § 1.1).

• The measurand characteristics are some of the most important aspects when choosing a measuring method. According to this, the measuring methods are divided as direct method, indirect method, differential method, ratio method, reciprocity method, substitution method and transfer method. However, a perfect measuring device does not exist and therefore aspects such as the measurement scope must be taken into account in order to reduce the errors and uncertainties up to the acceptable levels (see § 1.1.1).

• The measurement errors, which are always present in the measured quantity, are split between the device errors, the methodological errors, the human errors and the statistics. The result of the measurement is the sum of the true value, the systematic errors and the random errors. Systematic errors refer to the accuracy of the measurement, while random errors concern the precision of the measurement results. The measurement uncertainty is a characteristic of the result, and it reflects the repeatability of the measurement and its reproducibility (see § 1.1.2).

• The inspection of production and measuring equipment refers to the process of directly or indirectly evaluating by measurements the elements that from the system. The inspection procedure is usually performed according to certified guidelines, but the equipment manufacturer or the user tend to intervene with specific needs according to the device or the client's needs (see § 1.2).

After each inspection procedure, an inspection certificate is issued by the operator or the inspector. This offers an overview of the products rejected, corrected and accepted and facilitate further improvement. The measuring instrument chosen for the inspection must be calibrated with an instrument 10 times more precise (see § 1.2).

Even though the statistical inspection is still used for mass production, lately new inspection principles have been developed, such as flexible inspection system and on-machine inspections (see § 1.2).

• The calibration process refers to the entire procedure of evaluating a device, instrument or equipment in order to improve its accuracy. The calibration certificate is issued by an authorized person, and it depends on external factors such as usage and environmental conditions (see § 1.3).

The traceability of a measuring device has a crucial importance, and it is reached by following the guides of international standards of calibration (see \S 1.3).

• Any external factors variation from the standard procedure should be noted in the procedure reports. They may influence the final result and therefore they have to be monitored, controlled and recorded. Although usually the production and measuring environment cannot be entirely controlled, the accuracy and repeatability of the machines depend on it. Correction coefficients are applied in order to compensate the errors (see § 1.4).

• Temperature fluctuations are one of the most affecting environmental conditions. Punctual measurements with thermocouples, interferometers and infrared thermography are some of the measuring techniques used to identify heat sources since the temperature variations may contract or expand the materials of both the measuring equipment and the measurand.

The expansion coefficient of materials is the one that determines its stability along temperature variations therefore manifold standards suggests 20 °C as an optimal temperature for measurements. One of the steadiest materials at temperatures up to over 100° C is Invar (see § 1.5).

• Measuring instruments establish quantities or variables with respect to national and/or international standards (see § 2.1).

• The measuring instruments have various classifications, some of them being based on the output results, on the input data or precision (see § 2.2).

• The output results-based divides the measuring instruments into analogue, digital, display, recording, summing and integrated devices (see § 2.2.1).

• The input data splits the measuring devices into mechanical, electric, thermal, magnetic, optic and acoustic devices (see § 2.2.2).

• The classification based on precision, groups the devices into precision classes. Some of the most precise measuring devices are interferometers, autocollimators, electronic levers, reference encoders, etc. and they are usually required for production or measuring equipment calibration (see § 2.2.3).

• Interferometry was implemented in measurement procedures for the first time two centuries ago by Albert Michelson and it is up to this date considered one of the most accurate measurement devices, having steep precision, great compliance, high measuring speed and being a non-contact measurement (see § 2.3).

• Interferometry developed starting from the light wave's theories. Each wave is characterized by irradiance, phase angle, direction of propagation, wavelength, and direction of polarization. The beam intensity or its capacity to concentrate energy on a certain area during a limited amount of time is also known as irradiance. The angle determined by the displacement of a sinusoidal wave from a determined reference gives the phase angle. While the definition of direction of propagation is intuitive, when two waves have their only difference the direction of propagation, they create a pattern interference through their superimposition.

Wavelength is the length between the beginning and the end points of an oscillation cycle. Accuracy and repeatability of the interferometric measurements are, among others, given by the wavelength stabilization mechanism of the laser source. During an interferometric measurement, fringes are captured as outputs and its disturbance determines a displacement.

Since interferometry is based on a laser wave light wave emission for measurements, it may involve the Doppler Effect in various determinations, such as establishing the velocity of a moving target.

The direction of polarization is defined by the transverse direction of the electric field, and it can be either clockwise polarized or counterclockwise polarized (see § 2.3.1).

• The interferometry procedure requires two originally similar laser beams. In order to acquire them, one of the most used methods is by splitting the laser source beam through wavefront or amplitude division.

The wavefront division is achieved by means of screens with pinholes, mirrors, biprisms or lenses redirecting and dividing the light of a monochromatic point source to a common volume space where the beams create interference fringes through overlapping. Some of the most common wavefront division set-ups are the "Young's Double-Slit Experiment", Fresnel's biprisms, Fresnel's mirror, Billet's split lens and Lloyd's mirror.

The amplitude division is met when two or more light waves are obtained from the same wavefront. Through their recombination is achieved interferometry. Some of the most usual interferometric devices contain an amplitude division element such as beam splitters, diffraction graters or polarization prisms (see § 2.3.2).

• Various interferometers were developed along the time since Michaelson's interferometric set-up based on a white light directed through a beam splitter towards two different mirrors that redirect the beams towards a screen.

The Fizeau interferometer uses a laser source, a beam splitter, a camera and a reference surface in order to characterize an optical surface that only needs to be able to produce a clear interference pattern. In certain situations, the measuring may be captured simply by the eye of the operator.

In optical telecommunications and quantum mechanics applications, the Mach-Zehnder interferometer is manly used. It uses one light source, one detector, two mirrors and two beam splitters. Usually, the four optical elements are placed at a 45° angle as against the propagation direction.

The Sagnac interferometer is particularly stable and consists of the interference of two opposite direction beams directed by one beam splitter and two or more mirrors. It is particularly used in navigation and surveying systems, since it is designed to detect rotation.

Starting from Michaelson's interferometer, Rayleigh developed a model based on wavefront division method, producing two beam paths through two lenses, gas cells and compensators. It is used in various applications, from determination of the refractive indices of gasses to molecular weight measurements.

The Laser Tracker interferometer was firstly designed in 1985 and it was developed along the time to be used for CNC machines and MMCs volumetric calibration. The geometric errors are identified after at least 4 different positions of the laser source measurements where the reflector is tracked by the beam along a defined route (see § 2.3.3).

• Interferometry's measurement uncertainty is influenced by the environmental, geometric and other errors. The environmental errors depend on the light wavelength sensibility and the thermal expansion coefficients of the engaged materials. A wavelength compensation number is used to avoid the light wavelength error, while, for a minimum thermal error, a steady temperature and a pre-process preparation is necessary.

Dead path error is a geometric error that is related to the environmental conditions, but it is defined by the distance between the beam splitter and the point zero of the measurement.

In order to avoid it, environmental conditions steadiness and a distance minimization between the beam splitter and the beginning point of the measurement are required. When the laser beam is not perfectly aligned to the machine axis to be measured, an angle is formed that generates the cosine error. A better alignment leads to a smaller cosine error.

Another angle that creates an error may be generated by the reflector inclination due to imperfect movement. This error is almost impossible to be estimated or avoided (see § 2.3.4).

• Interferometry has a multitude of applications and it reformed metrology. Its typical set-up involves a laser beam source, beam splitters, mirrors, optic mounting kits and other supports and optics depending on the measurement type.

Even though interferometric measurements can be time consuming due to its set ups, alignment and realignment procedures, it is still widely used for calibration as a result of its accuracy and reliability.

By the instrumentality of interferometry, various properties and errors of production and measuring machines may be determined regarding positioning, straightness, flatness, squareness and parallelism.

The linear positioning measurement has a dedicated standard that describes all the procedure. This measurement implies a beam splitter, two reflectors and, when needed, guiding mirrors. One of the reflectors is positioned on a moving along the axis component of the machine and the other one is used for the reference beam.

The current set-up usually implies that one of optics is mounted in the main spindle. This aspect may considerably limit the measurable area.

The angular positioning measurement evaluates the angular positioning accuracy of the components. For the angular positioning measurements along a linear axis, one angular splitter and one angular reflector are used as optical elements to perform the procedure of this measurement type. The angular reflector is positioned on a part that moves along the axis to be evaluated. The differences of the two laser beams that return from the angular reflector indicates the angular error.

The guideways of a machine tool are evaluated through straightness measurements. It involves an angle mirror and usually a Wollaston prism.

The specific deviations associated to tables and flat surfaces are identified with a flatness measurement that implies measurements along two axes. The optical elements have to follow a predefined trajectory, based on Moody method, Grid map method or Half-grid map method.

Squareness and parallelism measurements are a combination of the previous optical elements and specific optical, fixing and gripping elements. While squareness measurement evaluates the perpendicularity between two axes, with linear movement, parallelism measurement evaluates this position between either two linear axes designed to be co-axial or between the axes of two rotational heads (see § 2.3.5).

• The interferometer manufacturers usually provide a mounting and fixing kit according to its applications, being designed for interchangeability when possible. The metallic parts from the kit are made from magnetic or non-magnetic stainless steel (see § 2.3.6).

• Even though they are still being studied, lattice structures began being used in increasingly more domains and even in mounting and gripping systems. Additive manufacturing and Design programmes facilitate the conception and achievement of these structures. One of the best materials for mounting and fixing systems is Invar due to its thermal stability, but Inconel718, TiAl6V4 and CoCrSP2 superalloy are also good options due to their characteristics (see § 2.3.7).

Part II.

Contributions to defining and constructive development of laser interferometry systems structure

Chapter 4. The directions, main objective and methodology of research and development on laser interferometry systems structure

4.1. Directions of research and development

Based on the data and conclusions drawn from the present state of the art analysis, it is assessed to be of perspective, for the laser interferometry systems, the research and development directions on:

- general structure and characteristics;
- interacting processes and analytical modelling;
- operational characteristics, in order to develop applications in active industrial environments;
- analysis of the lattice structure integration into measuring systems;
- specific databases.

4.2. The main objective of research and development activity

Taking into account the present state of the art and the above research and development directions, it is assumed as main objective of the present foremost research and development doctoral activity: **structure and attributes development of laser interferometry systems.**

4.3. Methodology of research and development

The research and development methodology are elaborated as a reference base for the achievement of the main objective of the present doctoral thesis, respectively, of the following chapters of the doctoral thesis, as well as for future research.

The structure of laser interferometry systems should be configured in correlation with characteristics of the measurements objective.

The current structure of the laser interferometry systems contains a holding system for the beam splitter assembly that obstructs the linear measurements capacity sometimes for more than 30% for machining centres.

Thus, an innovative structure of a laser interferometry system for linear position aims at providing full working space availability for measurements at similar precision.

An innovative structure of a laser interferometry system for linear position is to be conceptualized and, more specifically, research, detailed design, manufacture, testing and validation of an innovative splitter positioning device should be developed.

The innovative system should be able to be used for various types of laser interferometric systems and measurements. Therefore, testing and validation of the system simultaneously in different setups should be performed.

The main research and development stages should be as follows.

(A) Conceptualization of a new splitter positioning device

- Initial data analysis
- Minimum requirements determination
- Structure optimization conditions

The conceptualization starts from initial input of requirements and necessities. An elaborate analysis of the actual structure and attributes of laser interferometry systems must be accomplished in order to assure consideration of all design constraints.

The conceptualization should take into account, other than enlarging the measuring area, the materials choice due to its requirements and also, to consider the measuring procedures that can be performed by means of the new splitter positioning device. As it is well known, each measuring procedure requires different optical elements, so that, each of the linear, angular, rotational, etc. procedure should be closely analysed in order to understand their limits (see § 2.3.5).

There must be utilised the relevant results of the studies unrolled on the developed system devices.

The various types of measurands that are taken into account must be well determined and analysed.

Some of the most important elements to be established are:

- minimum dimension of functional space for the new holding element;
- evaluating of external loads;
- determination of the optimum holding area;
- configuring of the objective functions;
- determination of response constraints.

(B) Design of the splitter positioning device

In order to design a functional splitter positioning device, the concept has to be taken through the following steps:

- computer-aided design;
- finite element method analysis;
- structure optimization.

Starting from the conceptualization output, an initial iteration can be designed and finite element analysed. Various CAD systems can be used for this study part, such as SolidWorks, AutoCAD 3D, Catia or Ansys.

For the lattice structure optimization, an Ansys Topological Lattice Optimization Analysis can be performed. The geometry is to be built as an assembly in order to segregate the lattice area.

The assembly should be subjected to a static structural analyse prior to the structural optimisation. The optimized version of the part is given by the objective function and the response constraints that are to be chosen. The digital validation connects the initial data from the finite element analysis with the modified optimized structure and offers a similar analysis of the structure with the lattice structure acquired.

The topological optimization will be unrolled, as a structural optimization that adopts the material data, geometry and model from the static structural analysis and requests the objectives of this procedure along with the lattice type, density and cell size.

(C) Splitter positioning device manufacturing

For initial evaluation of the design, a polyamide fabrication can be completed due to its low production costs and high precision. This procedure and material do not require supports or a printing base. The procedure will consist of:

- pre-process: powder preparation, CAD model design, model orientation in virtual printing space and model slicing;
- manufacture: layer by layer selective laser sintering;
- post-process, as cooling, surface finishing through not sintered powder removal.

Depending on the final design, the SPD can be fabricated through conventional procedures or through additive manufacturing out of the chosen metal powder, function to characteristic's evaluation and its availability. The additive manufacturing process might require a more extensive preparation but it is more suitable for prototypes or single part fabrication.

The product design must be altered in order to fulfil particular additive manufacturing requirements. This manufacturing procedure will consist of:

- pre-process, as powder and printing base preparation, CAD model design according to additive manufacturing process requirements, model orientation in virtual printing space, supports generation and model slicing;
- manufacture, as layer by layer;
- post-process, as cooling, powder removal, stress relieving heat treatment, part removal from the printing base, supports removal from the part, surface polishing or grinding, hole and thread execution.

(D) Splitter positioning device validation

In order to assess the concept, design and manufacturing achievements, the new set-up for laser interferometry system is to be tested and validated.

Choosing the proper evaluation methods and conditions has a great impact over the reliability of this product.

Preliminary results can determine whether or not the new set-up results are to be taken into consideration. As a consequence, consecutive measurements with an existing set-up and the new set-up should be taken into consideration. This procedure offers the freedom of complete measurements for each set-up which fully examines their capabilities, while immediate measurement as recommended by the interferometer's manufacturer offers a reference base for the results.

A peer assessment of the two set-ups is given by measurements performed in parallel with two similar laser interferometry systems.

The environmental conditions have a significant influence over the interferometer's measuring results, so that, the measuring should be made in different working conditions.

The ISO 230-2 procedure, as an internationally approved method, is to be used in order to validate the results.

In summary, this methodology involves high tech procedures, some still in the development stage - lattice optimizations, 3D printing -, which require continuous testing during the various phases of product development. Equally, the development procedure does not present possibilities of simplification at the current stage of research but, once a first set of data is obtained, it is possible to follow the elimination of some steps, such as preliminary printing on plastic, or such intense experimental validation, etc.

Chapter 5. Structure and characteristics development of laser interferometry systems

The development of structure and characteristics of laser interferometry systems, including the author's published elements [G04], is as follows.

The present proposed structure and characteristics of laser interferometric systems are developed at high level of generality, open to particularizations, in context, as well to other description levels.

Let measurement object be a set, G, of geometric entities, G_i , $i = \overline{1, n}$,

$$G = \{G_i | i = \overline{1, n}\} \iff G = G_1 \cup G_2 \cup ... \cup G_i \cup ... \cup G_n$$
(5.1)

It is to underline that each geometric entity G_i is well-defined by the specific features - nominal characteristic, tolerance field, reference elements/ coordinate systems.



Fig. 5.1. A general structure of a laser interferometric system

The laser interferometric system, LIS, for measurements upon the geometric group G $(G = \{G_i | i = \overline{1, n}\})$, is structured so that to include elements as coordinate system $(CS_j, j = \overline{1, m})$, laser sources $(L_a, a = \overline{1, r})$, laser beams $(B_a, a = \overline{1, r})$, optical modules $(O_a, a = \overline{1, r})$, output beams $(B_a^*, a = \overline{1, r})$, detectors $(D_a, a = \overline{1, r})$, environmental sensors $(E_a, a = \overline{1, r})$ and compensation modules $(C_a, a = \overline{1, r})$, process software and computers, etc. as presented in Fig. 5.1.

The real constructive structure and characteristics of LIS depend on the measurement object features and other techno-economic conditions.

Thus, a *laser source* element and a *detector* (*receiver*) element are distinct system components or incorporated together in a *laser unit* component, optical elements type *beam splitter* and *interferometer* are distinct or - incorporated together in a *beam splitter*, some *environmental sensors* are included in a *compensation module*, etc.

Chapter 6. Development of an innovative laser interferometry system for linear position

The development of an innovative system for interferometric measurements, including the author's published elements [M09, M13, O01, S14], is as follows.

The optical elements of laser interferometry systems are, typically, fixed with magnetic V-blocks to the main parts of the machine tool, as seen in Fig. 6.1. The laser head, which is also the beam source and the signal receptor, is situated at a D_1 distance to the beam splitter assembly that is fixed through a magnetic V-block to the main spindle. The distance D_2 between the beam splitter and the retroreflector marks a safety distance between the two components to avoid collision. The retroreflector is connected through another magnetic V-block to the working table. From that point on, starts the measuring area (D_3) that ends at the travel limit of the working table on the linear guides on Y axis. In the particular case study with a Makino D200Z 5-axis vertical machining centre, the working table has a travel range of [-170, +130] mm, on Y-axis. The spindle, that has two linear guides, is able to travel on X and Z axes and intersects the working table in an area around the 0 point of the Y-axis.



Fig. 6.1. Regular set-up for a linear interferometric measurement on a CNC vertical machining centre

In a system structure as represented in Fig. 6.1, D_3 is limited to around 60% of the working table travel.

The main purpose of this development would be to enable the interferometric measurement of the whole working area.

In order to be able to access the whole working area of the axis travel, the new structure of laser interferometry systems is developed so that the beam splitter and retroreflector would not be gripped to elements whose axes intersect.

The two set-ups have a main difference which consists in the position of the beam splitter assembly.

The new structure gives the operator the possibility to evaluate the whole working area of the Y axis, fact that can be vital in machining centres to determine clearances and precision.

The general characteristics of a device that allows interferometric measure of the whole working area, SPD, should be:

• minimum mass;

• high rigidity (low compliance), related to the action of its weight and the weight of the supported parts (optical and additional elements);

• simple and reliable interlocking within a laser interferometry system.

Evaluating two of the commercial laser interferometry systems [R09, R10, D03], (§ 2.3.6), the dimensional characteristics required for the considered device have been determined.

The supported parts have a total mass of 560 g. To ensure a rigidity of SPD of high value, a calculus (pressure) force of 15 N was chosen, which is related to a mass of approximately 1.5 kg.

The rigidity is influenced by the material characteristics, so that several studies were performed to determine the type of material and lattice structure that can be used [M09, M13, S14, O01] (also, see § 2.3.7).

The splitter positioning device, SPD, is developed as 2 main different constructive types.

The finite element analysis of the steel part reveals a good rigidity having micrometric displacement, but its considerable mass of around 2.1 kg, and manufacturing challenges opens the path for further improvements.

The effective structure has been generated and multiplied using SolidWorks [D06] part features.

For coarse fittings and analysis, the finished part is subjected to a FEA and, thereafter, it is SLS fabricated out of PA2200 with EOS FORMIGA P110 additive manufacturing mechanism. The polyamide material PA2200 is chosen for this prototype, because it is a much cheaper and faster procedure compared to the metal additive fabrication.

A visual inspection indicates a small gap generated, most probably by incorrect handling; therefore, it can be taken into consideration that the current outer areas of structure are sensitive and easily deformable.

A further iteration of the device is corner rounded, in order to prevent deformation. The bottom of the structure is designed at the same level so that it can be placed on the metal printing plate without additional supports and for ease of removal from the support plate.

The specific parts of the device are SLS manufactured out of polyamide material PA2200 and subjected to testing in order to assess its applicability. An experimental testing on the prototype shows that the measurement range is significantly large (300 mm) in the measurements using the new device, comparing to the measurements without it. This important result reveals the utility of the splitter positioning device to enlarge the measurement area to the entire working area of the CNC centre. But, in the same time, the *errors*, generated because of the polyamide material in the constructive structure of the new device, are of high values, not acceptable, therefore, the next developments must take into consideration these elements.

Another iteration of the device is developed with its central body of lattice structure, corner rounded, and its dimensions were decreased as much as possible, using the advantages offered by the direct metal laser sintering process.

The DMLS process involves particular attention to the part details such as holes, edges, overhangs and base, therefore, the part is adjusted, with tear drop shape holes, angles optimization, same level base plan, no overhangs and material consumption minimization.

The positioning tuning system area undergoes specific changes

In order to avoid a possible misalignment caused by the structure deformation, a push system with two screws is designed to adjust the small angle movements.

The lattice generation is made by means of the Ansys [A12] lattice topological optimization module, taking into consideration the FEA output, the objective function and response constraints of interest, and the influential aspect that printed part should allow elimination of the loose powder (no embedded powder volumes).

The yielding of the part can be disregarded as it results from the von-Misses stress analysis. The stress differences between the two parts (full body and lattice body) were almost unchanged and the tension level is low enough that it does not need to be considered.

For this study, the cell dimension was of 5 mm. The optimization objective was minimization mass and compliance.

The density range was chosen in order to avoid embedded powder volumes and to not reach the minimum manufacturing size.

The body geometry to be optimized generated a high lattice density closer to the tripod grip and a low density at the other extremity.

Ansys produces an equivalent density map through this optimization therefore, the final step of the structural optimization resumes the previous static structural FEA but with the new geometry. The displacement differences between the two are micrometric.

Another iteration of the new device contains elements that allows the beam-to-beam splitter alignment, without the position tunning system.

In connection with the metal printing of the specific components of the device and DMLS fabrication process, printing supports of about 4 mm were added to the orientation holes and to the overall part in order to facilitate the wire cutting part removal from the printing plate during post DMLS process.

For this final geometry and the CoCr material, the Ansys programme determined an expected mass of approximately 500 grams, with almost 25 % less than the full body part, but at the same time, with 76% less than the mass of the first iteration. The percentage can change according to the density range, lattice cell dimensions, external loads and initial geometries.

For the pre-processing of the part, which is prior to the fabrication of the specific components of the device, the CAD assembly is exported to .stl format for additional part editing, support addition, problematic areas detection and machine definition in Magics [M14] programme. The parts are then exported as a cli. extension for the EPHatch [E07] programme to be sliced. In this case, the parts were sliced with a 0.03 mm increment, thus defining each of the 1566 layers to be printed. When the final parts are prepared, they are transferred to the printing equipment with an .epi extension.

The printing equipment used in this fabrication is a Shining 3D EP M250 3D model. Prior to the fabrication beginning, the powder is sieved, dried and deposited into the feed cartridge.

The additive manufacturing of the specific parts of the device would last under normal conditions for around 34 hours. In this particular case, the printing procedure finished after 240 hours, due to unknown cause.

During the post-processing procedure, after additive manufacturing, a cooling time is required before the building platform extraction. Afterwards, the powder is carefully removed from all the building platform and parts areas. The building platform is then removed and introduced in the sanding machine to smooth the surfaces and blow the remaining embedded powder.

The following step involves the stress relieving process. The powder manufacturer recommends a stress relieving procedure into the furnace, under argon atmosphere. The available equipment did not offer a controlled atmosphere option, therefore an adaptation of this procedure to non-controlled atmosphere was determined. The parts are initially heated up to 450°C for an hour, the temperature is maintained there for an hour, and then it is raised up to 800°C for one hour and it is maintained there for another one hour. Afterwards, the heat is turned off and the cooling process comes naturally.

After the relieving process, the material hardness may be determined. In this case, the parts capacity to resist deformation is of 44,5 HRC. This result falls within the margin estimated by the powder manufacturer, with their recommended stress relieving procedure.

The following steps regard parts extraction from the building platform and the needed machining after the printing. The parts extraction is made by means of wire electrical discharge machining.

The resulted parts are then post processed to remove sharp edges and the remaining supports. Several surfaces of the splitter support are grinded according to the requirements. Also, applying EDM procedures, the cylindrical hole and the associated thread were machined with specific electrodes.

Chapter 7. Validation of the new splitter positioning device within laser interferometry systems for linear position

The validation of the new splitter positioning device within laser interferometry systems for linear position, including the author's published elements [A04, S15], is as follows.

7.1. Experimental protocol and environmental conditions

As presented in the previous chapters, performing linear measurements with laser interferometry systems offers reliability and high accuracy but it also has certain constraints. One such constraint is determined by the placement of the optical elements, which requires a rigid metal area for the magnetic V-block. It often leads to a limitation of the measuring range for all the machining centres.

7.1.1. Experimental protocol

SPD is to be tested on three CNC centres, CNC Makino D200Z 5-axis machining centre, CNC Spinner VC1020 machining centre and CNC Spinner MC1020 machining centre.

During each test, two major laser interferometry systems, LISs, are configured and used, consecutively or in parallel (simultaneously), i.e.:

• a simple (regular) laser interferometry system "simple LIS",

• a laser interferometry system including SPD "LIS including SPD".

The considered LISs are structured with elements from Renishaw XL-80 laser systems.

The measurements procedure to be used is based on the elements described by ISO 230-2.

The environmental conditions for the measurements must be adapted to the equipment manufacturer's specifications. The standard suggests an optimal temperature of 20° C. But, even

under this situation, errors can occur and therefore the temperature value must always be mentioned in the test report.

The considered procedure is to be used for consecutive and parallel measurements. Various comparative measurements are unrolled with labels as presented in Fig. 7.7.

The results are displayed in graphs where the measuring points coordinates (position distances) along the machines travel Y axis are represented on the abscissa, and the corresponding positioning errors – on the ordinate.



Fig. 7.7. Measurement nomenclature

7.1.2. Study on environmental conditions

The effects of LIS environmental compensation are put into perspective, through two types of tests: one with environmental compensation unit and one without it. Preliminary studies were performed to determine the thermal influence on interferometric measurements [A04]. However, as the manufacturer states [R10], the XC-80 compensator unit can compensate the environmental conditions influence.

Two standard measurements were performed on the CNC Makino D200Z 5-axis vertical machining centre, and the results are represented in Fig. 7.9. Along the abscissa, the Y axis specific range values are kept within [-170, +10] mm range.





It is an obvious difference between the values given by the measurement with compensation, where the considered CNC centre reveals close to no positioning error, and the values given by the measurement without compensation, where the error goes up to over 12 μ m.

The specific analysis statistical indicators show the followings:

• the positioning error, the systematic positioning error and the range of mean bidirectional positioning error are strongly influenced by the environmental conditions;

• the positioning repeatability is of an acceptable value $(1.5 - 2.1 \,\mu\text{m})$, in each measurement case.

It is to be mentioned that for the considered measurement cases, the environmental conditions varied with less than 0.5° C for the temperature, while the other conditions were stable.

7.2. Consecutive tests in laboratory conditions

The consecutive tests are referring to the use of the LIS including SPD and, consecutively, the simple LIS.

The CNC machining centre for these tests was the CNC D200Z 5-axis vertical machining centre.

The laboratory conditions were considered to be in a closed environment, with limited traffic. All the environmental conditions were noted, and the XC-80 compensator was enabled.

Several groups of Y axis accuracy and repeatability of positioning measurements were conducted. From these, three groups (with order number 02, 03 and 06) of results are presented in Fig. 7.10. The graphs show the positioning errors along the Y axis. Along the abscissa, the Y axis points coordinates are limited to a [-170, +10] mm range when the simple LIS is used (see § 7.1), while the Y axis points coordinates are in the extended range of [-170, +130] mm when the LIS including SPD is used.

It can be noticed that the measuring cycles start and end up at approximately 0 μ m error, while along the evaluated Y axis (abscissa axis), it tends to increase. The SPD_02_c and the Simple_02_c measurements results have an increasing difference between similar points with a reach of about 4 μ m at the comparative point of 10 mm position coordinate. The other two sets of measurements, even though they have different values from the previous ones, they keep the same difference of around 4 μ m at the comparative point of 10 mm position coordinate.



Fig. 7.10. Y axis accuracy and repeatability of positioning measurements during consecutive tests

The specific analysis statistical indicators show the followings:

• generally, the values of each analysis statistical indicator at the measurements with the LIS including SPD and with the simple LIS are of the same order of magnitude;

• the positioning error, the systematic positioning error and the range of mean bidirectional positioning error at the measurements with the LIS including SPD are smaller or very close to those from the measurements with the simple LIS; the positioning repeatability value is of $1.8 - 4.9 \mu m$ at measurements with the LIS including SPD, and of $1.1 - 2.1 \mu m$ at measurements with the simple LIS.

The environmental compensation corrected the measurement errors that might have happened due to the environmental conditions' variation influence over the materials involved in the procedure or the laser wave. Even though the procedures were performed in a closed chamber, with little to no traffic, and only one operator was present. The air and material temperatures were quite different even though the consecutive sets of measurements were performed with minimum time intervals. The overall variations between the beginning and the end of a set go even up to 2° C, which might have changed the wavelength and considerably influence the measurements' results. The difference between air and material temperature is almost 1° C for all the sets except the first one, where the biggest temperature variation is also captured. Altogether, the temperatures reach up to 26° C, representing a variation of up to 6° C to the ideal temperature value of 20° C, recommended by the standards.

Overall, these measurements prove the applicability of SPD. It is obvious that an innovative device to support the beam splitter, enlarges the measuring range making possible the evaluation of the whole axis travel. As determined, a whole area of around 40% of the axis travel was added to the measuring range, using a LIS including SPD.

7.3. Parallel tests in laboratory conditions

The parallel tests are referring to the use simultaneously one simple LIS and one LIS including SPD.

The main feature of the parallel tests is that both different LISs are functioning in the same technical conditions associated to the machine working elements and in the same environmental conditions.

The working conditions are identical or similar to those presented in the previous subchapter: CNC D200Z 5-axis vertical machining centre, the general structure of each laser interferometry system, environmental sensors, XC-80 compensator, etc.

In particular, the simple LIS and the LIS including SPD. were configured using the specific elements from two LISs of different editions type Renishaw XL-80 (Fig. 7.12).
The set-up was particularly meticulous to be put in place, since the two laser sources have to be close next to each other so that each laser beam to reach the correspondent reflector. The beams can be aligned on the same time or separately.

It is underlined that, because of the simple LIS structure, during the considered parallel tests, the Y axis points coordinates are limited to the [-170, +10] mm range.

Several groups of Y axis accuracy and repeatability of positioning measurements were conducted. From these, three groups (with order number 05, 11 and 13) of results are presented in Fig. 7.14.

The values keep the pattern from previous consecutive tests (see § 7.2) and have quite the same differences between them of about 4-5 μ m at the maximum distance.



Fig. 7.14. Y axis accuracy and repeatability of positioning measurements during parallel tests

The specific analysis statistical indicators show the followings:

• generally, the values of each analysis statistical indicator associated to the LIS including SPD and to the simple LIS are of the same order of magnitude;

• the positioning error, the systematic positioning error and the range of mean bidirectional positioning error are smaller at measurements with the LIS including SPD, even 4 times smaller in some cases; the positioning repeatability value is of $1.1 - 3.4 \mu m$ at measurements with the LIS including SPD, and of $1 - 2.1 \mu m$ at measurements with the simple LIS.

The specific parameters were captured simultaneously by sensors of the two XC-80 compensation units. The parameters variations are not larger than 1° C at temperatures, 5.4 mbar at pressure and 7% at humidity, but if the XC-80 units would not implement the compensation, the wavelength would suffer changes.

7.4. Parallel tests in industrial conditions

The parallel tests in industrial conditions were performed on two different CNC machining centres, i.e., CNC Spinner VC 1020 machining centre - with table travel on axes X, Y and spindle travel on Z axis, as well as CNC Spinner MC 1020 machining centre – with spindle trave on all X, Y, Z axes.

The other working conditions are similar to those from parallel tests in laboratory conditions (see § 7.3), i.e.: simple LIS and the LIS including SPD were configured using the specific elements from two laser systems of different editions type Renishaw XL-80, both reflectors are placed on the same mounting pillar, etc.

(A) Parallel tests on CNC Spinner VC 1020 machining centre

It is to be noted that at CNC Spinner VC 1020 machining centre, the table Y axis working travel range is of 610 mm, while the maximum evaluated range was of about 480 mm because of the limit introduced by the simple LIS structure.

The results (Fig. 7.16) show a similar pattern, with a slight difference for the results associated to the LIS including SPD, of up to about maximum 15 μ m. The errors are under 10 μ m with small oscillations up until the point of 240 mm coordinate; from that point on, the error decreases until up to - 40 μ m for the simple LIS, and until up to - 20 μ m for the LIS including SPD. On the returning path, the values are slightly similar with a continuous difference of less than 10 μ m between the errors associated to the considered LISs.

The range of mean bidirectional positioning errors is of around 30 μ m for the LIS including SPD and of around 40 μ m for the simple LIS. The positioning repeatability is slightly similar between the two LISs with a small variation for the LIS including SPD. Overall, the positioning errors met in Y axis evaluation proved to be higher in the second part of the travel axis which probably of the wear level of the linear guides in that working area of the machine.



Fig. 7.16. Y axis accuracy and repeatability of positioning measurements during parallel tests on a CNC Spinner VC 1020 machining centre

The environmental conditions were stable. It can be noted that the difference between the air and the material temperature is about 2° C, similar to the laboratory onditions (see previous subchapter), but the overall temperature is higher than the recommended 20° C temperature, which indicates the need of the environmental compensator.

(B) Parallel tests on CNC Spinner MC 1020 machining centre

It is to be noted that at CNC Spinner MC 1020 machining centre, the spindle Y axis working travel range is of 580 mm, while the maximum evaluated range was of about 380 mm because of the limit introduced by the simple LIS structure.

The set-up for the simple LIS and LIS including SPD takes significant time and space.

The results follow a similar pattern of errors associated to the LIS including SPD and to the simple LIS, respectively, as presented in Fig. 7.19.



Part II. Contributions to defining and constructive

PhD Thesis

Summary

UPB

Fig. 7.19. Y axis accuracy and repeatability of positioning measurements during parallel tests on a CNC Spinner MC 1020 machining centre

The measurements start from the point of 0 coordinate of the Y axis; at the position of 40 mm along the axis, the positioning error is - $20 \,\mu m$. This error continues with small oscillations along the axis up to 120 mm position, where the error becomes -30 µm. The error with a very small decreasing continues up to the last value at 380 mm position. On the returning path, an abrupt error correction from around -30 µm to 0 µm is noticed. This is probably due to a clearance in the linear guides. From that point on, the returning points are within $\pm 10 \mu m$ error around the null value.

The bidirectional positioning errors are at around 40 µm while the positioning repeatability is slightly better for the simple LIS. A maximum value variation of about 15 µm can be noted between the two different LISs.

As in the previous industrial parallel tests, the environmental conditions were stable.

The comparative measurements, presented in this chapter, show that SPD, included in a LIS, meets the specific requirements. The experimental results reveal important similarities between the measurements performed with the LIS including SPD and the simple LIS, leading to an enhancement of the capabilities of LISs to evaluate the entire possible travel on certain axes of CNC centres.

Chapter 8. Final conclusions and main contributions regarding the structure and attributes development of laser interferometry systems

(1) From the analysis of the actual state of the art on measuring processes and systems, important conclusions have been drawn, as they are presented in the chapter 3.

(2) Based on the data and conclusions drawn from the present state of the art analysis, it has been assessed to be of perspective the research and development directions as presented in § 4.1.

(3) In relation to the present state of the art, as well as the research and development directions, it has been assumed as main objective of the foremost research and development doctoral activity (see § 4.2): structure and attributes development of laser interferometry systems.

(4) The relevant conclusions associated with the foremost research and development doctoral activity to achieve its main objective, guided by the methodological reference elements (see § 4.3), are as follows.

 \circ The development on general structure and characteristics of laser interferometry systems includes analytical descriptors and some general features referring to measurement object, coordinate system, lasers, laser beams, optics, output beams, detectors, environmental sensors and compensation modules, electronics, process software and computer(s), assembly, calibration and other elements (see § 5).

 \circ Due to the actual set-up of the optical elements, the laser interferometry systems for linear measurements are often restricted to a limited measuring area due to their placement inside the working area of the equipment. As an example, for a Makino D200Z 5-axis vertical machining centre, an evaluation of the positioning error of the Y axis can be performed on only around 60% of the working table travel. In order to overcome this limitation and to allow for a full range of measurement, a splitter positioning device, SPD, for laser interferometry system structures was developed as a new system component (see § 6. 1).

 \circ The general characteristics of SPD should be: minimum mass; high rigidity (low compliance), related to the action of its weight and the weight of the beam splitter assembly; simple and reliable interlocking within a laser interferometry system (see § 6. 2.1).

 \circ For the design phase, the load that the device was to bear was approximated to 15 N, taking into account its own weight and that of the parts involved. SPD should deform as little as possible under this load. The resulting rigidity of the component is a cumulated result of the part geometry, the fabrication method and material being used (see § 6.2.1).

 \circ SPD can be designed and manufactured in various forms and from different materials, but by taking into consideration the resulting mass and material homogeneity (see § 6.2.2).

The central part of the SPD body is analysed as solid, hollow or lattice form, made of metallic or polyamide material.

The load applied to SPD leads to bending induced deformation. In order to compensate for this inconvenience, a push system with two screws has been designed to adjust the small angle movements.

 \circ A SPD that integrates lattice structures obeys all design criteria, as it can be customized and its geometry can be optimized according requirements. The lattice structures, presently, can only be obtained by additive manufacturing. This implies adjusting the component design also in accordance with the fabrication type, since practice has shown that certain 3D printed geometries cannot faithfully follow the CAD design (see § 6.2.2).

• A proper fabrication method might be of use for a concept first evaluation, to ensure the feasibility of the geometry. A polymer-based SPD was firstly fabricated, using a PA2200 powder and an SLS machine. By analysis the quality of fabrication, it was determined that rounder corners are more suitable, along with material use optimization and other design improvements prior to DMLS fabrication. The resulting polymer prototype was also considered for use in interferometric measurement but, as it exhibited a viscoelastic behaviour (continuous deformation under load), it did not respect the stiffness design criteria and was eliminated from further scrutiny.

 \circ The lattice structure can be integrated into the SPD main part, with identical cell characteristics or through an optimization procedure, by considering the cell lattice type and the optimization objectives. Commercially available FEA programs have started in the recent years to supply topological optimisation modules or algorithms. Ansys software was used for SPD structure optimisation using lattice cell structures, geometry in accordance with a set of optimisation parameters and design constraints (see § 6.2.2).

 \circ The DMLS fabrication method requires special adjustments of the geometry, such as tear drop shape for the holes, supports for overhangs or angle changes for some supports, etc. to ensure the absence of fabrication induced defects. Due to part possible thermal deformations, in order to assure the prescribed precision of the SPD surfaces, the flange and boss positioning surfaces are designed to be machined after the parts are additive manufactured (see § 6.2.2).

 \circ Material analysis has led to conclude that the most suited one for the current application is CoCrSP2 superalloy, supplied in a powder form. In terms of pre-processing of the fabrication phase, both the geometry and the alloy powder have led to a layer thickness of 0,030 mm, resulting a total of 1566 layers (see § 6.3.1).

 \circ While, during the additive fabrication, additional issues risk occurring that determine irregular powder distribution, such as the recoater blade hitting certain rising areas already sintered, these events did not influence in a noticeable way the fabricated parts structure (see § 6.3.2).

 \circ The post-processing phase of the metallic SPD fabrication requires the building platform and parts extraction from the fabrication area, a stress relieving procedure, through a thermo-curing cycle and the parts removal from the building platform, usually performed with a wire EDM. Due to the material hardness, the boss threaded hole is recommended to be performed by EDM (see § 6.3.3).

 \circ The assembly can be easily integrated into the measuring set-up, but it can also suffer improvements to the holding area, improvements which are certain to appear with the extensive use of the SPD and increase of the number of laser interferometer models for which it is used (see § 7.1).

 \circ The ISO 230-2 standard was used. It describes the measuring procedure for CNC axes evaluation. (see § 7.1.1).

 \circ Validation of the new achieved splitter positioning device was performed by using a simple laser interferometry system and a laser interferometry system including the considered device for linear position, within consecutive tests and parallel tests, in laboratory and industrial conditions, with a continuous evaluation of environmental parameters (see § 7.2, 7.3 and 7.4).

The comparative measurements performed in this thesis have shown that SPD meets its requirements.

(5) To the achievement of the main objective of the research and development doctoral activity, the present PhD thesis brings several contributions, of which the most important are as follows.

• The profound analysis of the actual state of the art on measuring processes and systems, including interferometry and interferometers, additive manufacturing of grippers and mounting parts, lattice structures, as well as defining the main objective of the foremost research and development doctoral activity as structure and attributes development of laser interferometry systems.

• Development of defining analytical descriptors for general structure and characteristics of laser interferometry systems, referring to measurement object, coordinate system, lasers, laser beams, optics, output beams, detectors, environmental sensors and compensation modules, electronics, process software and computer(s), assembly, calibration, and other elements.

• Conceptualization, design and finite element simulation of a beam splitter new positioning device, based on creation and analysis of several different constructive variants, with optimization by integrating lattice structures to reduce its weight.

• Manufacturing of the splitter positioning device parts, by selective laser sintering of polyamide device variants, and by direct metal laser sintering of superalloy device variant, as well their preliminary evaluation.

• Validation of the new achieved splitter positioning device, by including it in a laser interferometry system structure for linear position, performing measurements within consecutive tests and parallel tests, in laboratory and industrial conditions, leading to development favourable results, but also regarding the generic calibration with laser interferometry systems.

* * *

The present PhD thesis, by problematic, approach and results, develops a series of studies, analysis, concepts, designs, parts and experimental results concerning the development of an innovative splitter positioning device to be included in laser interferometry systems.

The *scientific importance* of this thesis is sustained by its contributions to: development of defining analytical descriptors for general structure and characteristics of laser interferometry systems; conceptualization, design and finite element simulation of a beam splitter new positioning device, with optimization by integrating lattice structures; manufacturing of the splitter positioning device parts by direct metal laser sintering of a superalloy powder; validation of the new achieved splitter positioning device by performing measurements in laboratory and industrial conditions, leading to development favourable results.

The *practical importance* of this thesis consists in the fact that the achieved descriptors for general structure and characteristics of laser interferometry systems, part conceptualization, design, fabrication and validation on the beam splitter new positioning device are useful, by case, to individuals, students, academic staff, industry specialists, various educational and economic organizations, to increase the laser interferometry systems matters.

The problematics of laser interferometry systems, as potential perspective on interacting processes, analytical modelling and operational characteristics associated to laser interferometry systems, in order to develop their applications in active industrial environments, requires a complex, continuous and profound research and development activity.

PhD Thesis *Summary*

References

[A01] ***, ASRO - Organismul Național de Standardizare, https://www.asro.ro/despre-asro/ (accessed on 30.03.2021).

[A02] ***, ASRO - Organismul Național de Standardizare, Statutul Asociației de Standardizare din România - ASRO, updated on 15.10.2015.

[A03] Artimon (Stochioiu) F. P. G., Identification expérimentale du comportement viscoélastique d'un composite stratifié lin/époxy – dissertation thesis, Institut Superieur de l'Automobile et des Transports, Université de Bourgogne, 2016-2017.

[A04] Artimon (Stochioiu) F. P. G., Stochioiu C. and Popan G., *Thermal influence on positioning error and position repeatability of a machining center axes*, U.P.B. Scientific Bulletin, Series D, Vol. 83, Iss. 2, 2021, ISSN 1454-2358

[A05] ***, Agilent Technologies Inc., Agilent Laser and Optics – User's Manual, Vol. 1, Ed. 5, 2007, https://www.uzimex.cz/soubory/20080403 laser optika cast 1.pdf (accessed on 01.10.2021)

[A06] Azman A., Vignat F. and Villenueve F., *Evaluating current cad tools performances in the context of design for additive manufacturing*, Joint Conference on Mechanical, Design Engineering & Advanced Manufacturing, 2014, France

[A07] ***, Althen Sensors & Controls., PPV / PPV80 Mounting Plate - AISI 304 steel upper plate for foot mounting, https://www.althensensors.com/sensors/weighing-sensors-load-cells/weighing-modules-mounting-kits-for-load-cells/5645/ppvppv80-mounting-plate/ (accessed on 10.02.2022)

[A08] Ashby M. F., *Material selection in mechanical design*, 3rd Edition, 2005, Elsevier Butterworth Heinemann, ISBN 0 7506 6168 2

[A09] ***, Air Products and Chemicals Inc., *Additive Manufacturing and Gas Applications Technology*, https://www.airproducts.com/-/media/airproducts/files/en/330/330-19-001-us-additive-manufacturing-and-gas-applicationstechnology.pdf (accessed on 10.02.2022)

[A10] ***, ArcelorMittal, *Invar – Controlled expansion alloys*, https://www.spacematdb.com/spacemat/manudatasheets/Invar-AM.pdf (accessed on 10.02.2022)

[A11] ***, AZO Materials, *Stainless Steel - Grade 316L - Properties, Fabrication and Applications (UNS S31603)*, https://www.azom.com/article.aspx?ArticleID=2382 (accessed on 10.02.2022)

[A12] ***, Ansys® Academic Research Mechanical, Release 2021 R2

[B01] Bucher J. L., *The Metrology Handbook*, Second edition, ASQ Quality Press, Milwaukee, Wisconsin, 2012, ISBN 978-0-87389-838-6.

[B02] Bedson P. and Sargent M., *The development and application of guidance on equipment qualification of analytical instruments*, Accreditation and Quality Assurance vol. 1, Iss. 6, 1996, pp. 265-274, ISSN 09491775, DOI 10.1007/s007690050083.

[B03] Bryan J., Livermore L., *International Status of Thermal Error Research* (1990), CIRP Annals - Manufacturing Technology, Vol. 39, Iss. 2, pp 645-656, ISSN: 1726-0604, DOI: 10.1016/S0007-8506(07)63001-7.

[B04] Barron T. H. K. and White G. K., *Heat Capacity and Thermal Expansion at Low Temperatures*, Springer Science+Business Media New York, 1999, ISBN 978-1-4613-7126-7, DOI 10.1007/978-1-4615-4695-5.

[B05] Blakenbach K., *Introduction to Display Metrology*. In: Chen J., Cranton W., Fihn M. (eds) Handbook of Visual Display Technology. Springer, Berlin, Heidelberg, (2012), https://doi.org/10.1007/978-3-540-79567-4_142

[B06] Bunch B. and Hellemans A., The History of Science and Technology, Houghton Mifflin Company, 2004, ISBN 0-618-22123-9

[B07] ***, Basic Electromagnetic Wave Properties, <u>https://www.olympus-lifescience.com/es/microscope-</u> resource/primer/java/wavebasics/ (accessed on 25.08.2021) UPB PhD Thesis *Summary*

[B08] Bass M., *Handbook of Optics*, Vol.1: Geometrical and Physical Optics, Polarized Light, Components and Instruments, 3rd Edition, 2010, ISBN 978-0-07-162925-6

[B09] Briers J. D., *Optical testing: a review and tutorial for optical engineers*, Optics and Lasers in Engineering, Vol. 32, Iss. 2, Pp. 111-138, 1999, doi: 10.1016/S0143-8166(99)00062-7

[B10] Bünnagel R., Oehring H. A. and Steiner K., *Fizeau Interferometer for Measuring the Flatness of Optical Surfaces*, Applied Optics, Vol. 7, No. 2, pp. 331-336, 1968, doi: 10.1364/ao.7.000331

[B11] Bucher D., Richards E. G. and Brown W.D., *Modifications of the Rayleigh Interferometer in the ultracentrifuge for use with Heme and other light-absorbing proteins*, Analytical biochemistry, Vol. 36, pp. 368-380, 1970, doi: 10.1016/0003-2697(70)90372-6

[B12] Begović E., Plančić I., Ekinović S. and Ekinović E., *Laser Interferometry – Measurement and Calibration Method for Machine Tools*, 3rd Conference "MAINTENANCE 2014", 2014, PP. 19-28

[B13] Brzhezinskii M. L. and Simakhina G. V., *Interferometer of improved sensitivity for measuring small linear displacements*, Translated from Izmeritel'naya Tekhnika, No. 5, pp. 44-45, 1972

[C01] Chiarini A., Japanese total quality control, TQM, deming's system of profound knowledge, BPR, lean and six sigma: Comparison and discussion, International Journal of Lean Six Sigma, 2011, Vol. 2, Iss. 4, pp 332-355, DOI: 10.1108/2040146111189425.

[C02] Clifford P. M., *The international vocabulary of basic and general terms in metrology*, Measurement, 1985, Vol. 3, Iss. s, pp 72-76, DOI: 10.1016/0263-2241(85)90006-5.

[C03] Castrup H. T., Eicke W. G., Hayes J. L., Mark A. and Martin R. E., *Metrology: Calibration and measurement processes guidelines*, NASA Reference Publication 1342, June 1994, Pasadena, California.

[C04] Castro H. F. F., Uncertainty analysis of a laser calibration system for evaluating the positioning accuracy of a numerically controlled axis of coordinate measuring machines and machine tools, Precision Engineering, vol. 32, pag. 106-113, 2008, doi:10.1016/j.precisioneng.2007.05.001.

[C05] Cable M., Calibration: A Technician's Guide, The Instrumentation System and Automation Society. ISA, 2005.

[C06] Clark A. F., Low temperature thermal expansion of some metallic alloys, Cryogenics, vol. 8, iss. 5, pag. 282-289, 1968.

[C07] Clark A. F., Fujii G. and Ranney M. A., *The thermal expansion of several materials for superconducting magnets*, IEEE Transactions on Magnetics, vol. 17, iss. 5, pag. 2316-2319, 1981.

[C08] Clark A. F., Ekin J. W. and all, Materials at Low Temperatures, American Society for Metals, 1983, ISBN: 0-87170-146-4.

[C09] Chunlei J., Zihua Z. and Chengwei L., *Vibration measurement based on Multiple Self-Mixing Interferometry*, Optics Communications, Vol. 367, 2016, pp. 227-233, DOI: https://doi.org/10.1016/j.optcom.2016.01.032

[C10] Culshaw B., *The optical fibre Sagnac interferometer: An overview of its principles and applications*, Measurement Science and Technology, Vol. 17, Iss. 1, 2006, DOI: 10.1088/0957-0233/17/1/R01

[C11] Castro H. F. F. and Burdekin M., *Evaluation of the measurement uncertainty of a positional error calibrator based on a laser interferometer*, International Journal of Machine Tools & Manufacture, Vol. 45, 2005, Pp. 285-291, DOI: 10.1016/j.ijmachtools.2004.08.012

[C12] Canero-Nieto J. M., Spinola C. G., Moreno-Aranda G., Bonelo J. M. and Martin-Vasquez M., *Contactless real time measuring system for the length of sheets in stainless steel production lines*, International Instrumentation and Measurement Technology Conference, Proceedings, Vol. 1, 2012, Pp. 2295-2299, DOI: 10.1109/I2MTC.2012.6229426

[C13] Chen J. R., Ho B. L., Lee H. W., Pan S. P. and Hsieh T. H., *Research on Geometric Errors Measurement of Machine Tools Using Auto-Tracking Laser Interferometer*, World Journal of Engineering and Technology, Vol. 6, 2018, Pp. 631-636, DOI: 10.4236/wjet.2018.63039

[D01] ***, Display devices, <u>https://electronics-club.com/display-devices/</u> (accessed on 26.05.2021)

[D02] Donges A. and Noll R., *Laser Measurement Technology - Fundamentals and Applications*, Springer Series in Optical Science – Vol. 188, 2015, ISBN 978-3-662-43633-2, DOI: 10.1007/978-3-662-43634-9

[D03] ***, Dual-Frequency Laser Interferometer ZLM 700 ZLM 800, <u>https://sciencedocbox.com/68820934-Space and Astronomy/Dual-frequency-laser-interferometer-zlm-700-zlm-800.html</u> (accessed on 10.02.2022)

[D04] Deshpande V. S., Fleck N. A. and Ashby M. F., *Effective properties of the octet-truss lattice material*, Journal of the Mechanics and Physics of Solids, © 2001 Elsevier Science Ltd, Vol. 49, Iss. 8, pp. 1747 – 1769, 2001, DOI: 10.1016/S0022-5096(01)00010-2

[D05] ***, Dimetix AG, Switzerland, Laser Distance Sensors – Starter kit accessories, <u>https://www.s-d-a.sk/dimetix/pdf/prislusenstvo.pdf</u> (accessed on 10.02.2022)

[D06] ***, DSS SolidWorks. Dassault Systems. Concord, MA: SolidWorks Corp., 2015.

[E01] ***, https://www.engineeringtoolbox.com/linear-expansion-coefficients-d_95.html (accessed on 12.02.2021).

[E02] Ellis J. D., Field Guide to Displacement Measuring Interferometry, SPIE Press, 2014, ISBN 978-0-8194-9800-7, DOI: 10.1117/3.1002328

[E03] Elmadih W., Leach R. and Syam W., University of Nottingham, Manufacturing Metrology Team, *Design of additively manufactured lattice structures for precision mechanical force loops*, <u>https://www.nottingham.ac.uk/research/groups/advanced-manufacturing-technology-research-group/research/manufacturing-metrology-team/our-research/design-of-additively-manufactured-lattice-structures-for-precision-mechanical-force-loops.aspx, (accessed on 10.02.2022).</u>

[E04] ***, EOS GmbH, EOS Aluminium AlSi10Mg - Material Data Sheet, https://www.eos.info/03_system-related-assets/materialrelated-contents/metal-materials-and-examples/metal-material-datasheet/aluminium/material_datasheet_eos_aluminiumalsi10mg_en_web.pdf and https://fathommfg.com/wp-content/uploads/2020/11/EOS_Aluminium_AlSi10Mg_en.pdf (accessed on 10.02.2022).

[E05] ***, EOS GmbH, *EOS Titanium Ti64 - Material Data Sheet*, <u>http://www.harbec.com/wp-content/uploads/2018/11/ti-ti64_m270_material_data_sheet_12-08_en.pdf</u> and <u>https://www.eos.info/03_system-related-assets/material-related-contents/metal-materials-and examples/metal-material-datasheet/titan/ ti64/material_datasheet_eos_titanium_ti64_grade23_premium_en_web.pdf</u> (accessed on 10.02.2022).

[E06] ***, EOS GmbH, EOS CobaltChrome MP1- Material Data Sheet, http://www.e-manufacturing.it/downloads/EOS_CobaltChrome_MP1.pdf (accessed on 10.02.2022).

[E07] ***, Eplus3D Tech GmbH, EP-Hatch 2020

[F01] Fridman A. E., *The quality of measurements – A metrological reference*, translated by Sabak A. and Makinen P., Springer, 2012, ISBN 978-1-4614-1477-3, DOI 10.1007/978-1-4614-1478-0.

[F02] Figliola R. S. and Beasley D. E., Theory and Design for Mechanical Measurements, Fifth Edition, 2011 John Wiley & Sons.

[F03] ***, Feanor R&D Metrology, Special Tools, Q-Software, *Principles of Operation*, <u>http://www.feanor.com/laser_principles.pdf</u> (accessed on 14.10.2021).

[F04] Fan J., Tao H., Wu C., Pan R., Tang Y. and Li Z., *Kinematic errors prediction for multi-axis machine tools' guideways based* on tolerance, International Journal of Advanced Manufacturing Technology, 2018, Vol. 98, Iss. 5-8, Pp. 1131-1144, DOI 10.1007/s00170-018-2335-9

[G01] Grabe M., *Measurement Uncertainities in Science and Technology*, 2nd Edition Springer-Verlag Berlin Heidelberg 2005, 2014, ISBN 978-3-319-04887-1, DOI 10.1007/978-3-319-04888-8.

[G02] Gupta S. V., *Measurement uncertainties – Physical parameters and calibration of instruments*, Springer-Verlag Berlin Heidelberg, 2012, ISBN 978-3-642-20988-8, DOI 10.1007/978-3-642-20989-5.

[G03] Grous A., *Applied Metrology for Manufacturing Engineering*, ISTE Ltd and John Wiley & Sons, Inc., 2011, ISBN 978-1-84821-188-9. [G04] Gheorghe M., Stochioiu F. P. G., Manolache D., Dijmărescu M. R. and Iliescu D., *Analysis and development on general structure and characteristics of laser interferometry systems*, IOP Conf. Ser.: Mater. Sci. Eng. 1268 012009, 2022, DOI 10.1088/1757-899X/1268/1/012009.

[H01] Hariharan P., Basics of Interferometry, Academic Press, Sydney, Australia, 2013, 2nd Edition, 2007, ISBN 978-0-12-373589-8.

[H02] Hecht E., Optics, Global Edition, 5th Edition, Adelphi University, Pearson Education Limited, 2017, ISBN: 978-1-292-09693-3.

[H03] Huang Y., Yang Y., Liang J., Miao Z., Zhao M. and Zheng Y., *An optical glass plane angle measuring system with photoelectric autocollimator*, Nanotechnology and Precision Engineering, Vol. 2, Iss. 2, Pp. 71-76, DOI: 10.1016/j.npe.2019.06.001

[H04] ***, High Temp Metals, *INCONEL* 718 *TECHNICAL DATA*, https://www.hightempmetals.com/techdata/hitempInconel718data.php (accessed on 10.02.2022).

[I01] Ishikawa K., *Guide to Quality Control*, Industrial Engineering and Technology, Asian Productivity Organization, Hong Kong, 1976, ISBN: 92 833 1035 7.

[I02] Iliescu D., Dijmărescu M. R. and Gheorghe M., *Configuration management by dolce upper-level ontology*, Conference Paper in Academic Journal of Manufacturing Engineering, 2015, Vol. 13, Iss. 2, ISSN: 15837904.

[I03] ***, ISO 9000 - WHAT IS THE ISO 9000 STANDARDS SERIES?, https://asq.org/quality-resources/iso-9000 (accessed on 27.05.2020).

[I04] ***, ISO 9000 Family Quality Management, https://www.iso.org/iso-9001-quality-management.html (accessed on 27.08.2020).

[I05] ***, International Organization for Standardization, International Classification for Standards, ICS edition 7, 2015.

[I06] Iliescu C., Ionescu-Golovanov C., Szabo W., Szekely I. and Bărbulescu D., Măsurări electrice și electronice, EDP, Bucharest, 1983.

[I07] ***, Institutul de Lingvistică "Iorgu Iordan" al Academiei Române - *Dicționarul explicativ al limbii române*, Editura Univers Enciclopedic, 1998.

[I08] ***, ISO/IEC 17025:2017, General requirements for the competence of testing and calibration laboratories, 3rd Edition.

[I09] ***, ISO 14644-1:2015, *Cleanrooms and associated controlled environments* — *Part 1: Classification of air cleanliness by particle concentration*, 2nd Edition.

[I10] ***, ISO 230-2:2014, Test code for machine tools — Part 2: Determination of accuracy and repeatability of positioning of numerically controlled axes, 4th Edition.

[I11] ***, Interference of waves, https://www.fizzics.org/interference-of-waves/ (accessed on 21.09.2021)

[I12] Iftimia N. V., Hammer D. X., Bigelow C. E., Rosen D. I., Ustun T. et al., *Toward noninvasive measurement of blood hematocrit using spectral domain low coherence interferometry and retinal tracking*, Optics Express, Vol. 14, Iss. 8, 2006, pp. 3377-3388, DOI: https://doi.org/10.1364/OE.14.003377

[I13] ***, ISO 230-1:2012, Test code for machine tools — Part 1: Geometric accuracy of machines operating under no-load or quasi-static conditions, 3rd Edition.

[J01] Jang Y. S., Wang G., Hyun S. et al., *Comb-referenced laser distance interferometer for industrial nanotechnology*, Scientific Reports, Vol. 6, 2016, DOI: https://doi.org/10.1038/srep31770.

[J02] Jozwik J., Jacniacka E. and Ostrowski D., *Estimation of uncertainty of laser interferometer measurement in industrial robot accuracy tests*, ITM Web of Conferences, Vol. 15, 2017, DOI: 10.1051/itmconf/20171504005.

[J03] Jia M., Zhang J., Gao R., Zhao D. and Peng T., *Precision measurement of squareness of large rectangular square*, Ninth International Symposium on Precision Engineering Measurement and Instrumentation, Vol. 9946, 2015, DOI: 10.1117/12.2180699.

[K01] Kutz M., *Handbook of measurement in science and engineering – Volume 1*, John Wiley & Sons, Inc., Hoboken, New Jersey, 2013, ISBN 978-0-470-40477-5.

[K02] Knapp W., *Measurement Uncertainty and Machine Tool Testing*, CIRP Annals - Manufacturing Technology, Vol. 51, Iss. 1, pp. 459 – 462, 2002, ISSN 0007-8506.

[K03] Kirby R. K., Hahn T. A. and Rothrock B. D., *Thermal Expansion*, The American Institute of Physics Handbook, 3rd edition, McGraw-Hill Book Company, New York, 1972, Chapter 4f, ISBN 07-001485-X.

[K04] Kartashev A. I. and Orshanskii D. L., *Classification of measuring devices by their accuracy*, Measurement Techniques, Iss. 5, pp. 447–451, 1962, https://doi.org/10.1007/BF00986694.

[K05] Kawamura S., Ando M., Nakamura T., Tsubono K., Tanaka T., Funaki I et al., *The Japanese space gravitational wave antenna* - *DECIGO*, Journal of Physics: Conference Series, Vol. 122, 2008

[K06] Kumar Y. and Chatterjee., Simultaneous measurement of refractive index and wedge angle of optical windows using Fizeau interferometry and a cyclic path optical configuration, Applied Optics, Vol. 48, 4756-4761, 2009, doi: 10.1364/AO.48.004756

[L01] Leach R. K. and Smith S. T., Basics of precision engineering, CRC Press, Taylor & Francis Group, 2018, ISBN 9781498760874.

[L02] Lo C. H., Yuan J. and Ni J., *An Application Of Real-Time Error Compensation On A Turning Center*, International Journal of Machine Tools and Manufacture, vol. 35, no. 12, pp. 1669-1682, UK, 1995.

[L03] Lahiri A., Basic Optics, Elsevier Inc., 2016, doi: 10.1016/c2015-0-04613-5.

[L04] Lau K., Hocken R. and Haynes L., *Robot performance measurements using automatic laser tracking techniques*, Robotics & Computer-Integrated Manufacturing, USA, Vol. 2, No. ³/₄, pp. 227-236, 1985

[L05] Lipus L. C., Budzyn G. and Acko B., *Analysis of Laser Interferometer Measurement Uncertainty by Simulating Error Sources*, International Journal of Simulation Modelling, Vol. 20, Iss. 2, Pp. 339-350, 2021, doi: 10.2507/IJSIMM20-2-563.

[L06] Loughridge R. and Abramovich D. Y., *A tutorial on laser interferometry for precision measurements*, Proceedings of the American Control Conference, Pp. 3686-3703, 2013, doi: 10.1109/acc.2013.6580402.

[L07] Leach R., *CIRP Encyclopaedia of Production Engineering*, Chapter: Abbe error/offset, pp. 1 – 4, 2015, DOI: 10.1007/978-3-642-35950-7_16793-1

[L08] Liu C. H., Jiwe W. Y., Hsu C. C. and Hsu T. H., *Development of a laser-based high-precision six-degrees-of-freedom motion* errors measuring system for linear stage, Review of Scientific Instruments, Vol. 76, Iss. 5, 2005, doi: 10.1063/1.1915520

[L09] Langenbeck P., *Interferometry for precision measurement*, Tutorial Texts in Optical Engineering, Vol. TT94, SPIE PRESS, USA, 2014, ISBN 978-0-8194-9142-8

[L10] Lacota S. and Gorog A., *Flatness measurement by multi-point methods and by scanning methods*, Journal of Interdisciplinary Research, 2011

[L11] Li Y., Gu H., Pavier M. and Coules H., *Compressive behaviours of octet-truss lattices*, Proceedings of the Institution of Mechanical Engineers - Part C: Journal of Mechanical Engineering Science, Vol. 234, Iss. 16, pp. 3257 – 3269, 2020, DOI: 10.1177/0954406220913586

[M01] Mittag H. -J., Rinne H., Statistical methods of quality assurance, Chapman & Hall, London, UK, ISBN 0 412 55980 3.

[M02] Morris A. S., Langari R., *Measurement and Instrumentation – Theory and Application*, Second edition, Academic Press, 2016, ISBN: 978-0-12-800884-3.

[M03] ***, Measurement System Basics – Analog vs. Digital, <u>https://www.keyence.com/ss/products/measure-sys/measurement-selection/basic/analog-digital.jsp</u> (accessed on 26.05.2021)

[M04] Mureșan A., *Aparate de măsură și control*, <u>https://biblioteca.regielive.ro/cursuri/agronomie/aparate-de-masura-si-control-221978.html</u> (accessed on 12.06.2021).

[M05] Maowei J., Xiaojing L., Shunchuan W., Yongtao G. and Linlin G., *Use of SAR interferometry for monitoring illegal mining activities: A case study at Xishimen Iron Ore Mine*, Mining Science and Technology, China, Vol. 21, 2011, pp. 781–786, DOI: 10.1016/j.mstc.2011.05.039

[M06] Micu C., Dodoc P., Diaconescu G. and Manolescu A., *Aparate si sisteme de masurare in constructia de masini*, Editura Tehnică, București, 1980.

[M07] Malacara D. and Malacara Z., *Chapter 29: Optical Metrology*, Handbook of Optics: Devices, Measurements, and Properties, Vol. 2, 2nd Ed., 1992

[M08] Modi B. S., George P. M. and Sheth S. M., *Study and Investigate Effect of Cutting Parameters on Flatness for Dual Plate Check Valve*, Conference: 7th international conference on Advanced computing & Communication Technologies, Vol. 4, 2013, DOI: 10.13140/2.1.3867.7286

[M09] Milodin N. L, Popa N. M., Tutoveanu M. and Artimon F. P. G., *Compression behaviour of PA2200 lattice structures*, International Journal of Mechatronics and Applied Mechanics, Vol. 1, Iss. 9, pp 192-200, 2021, DOI: 10.17683/IJOMAM/ISSUE9.28

[M10] ***, Makino GmbH, The Perfect Combination of Precision and Speed - The Makino D200Z, https://www.youtube.com/watch?v=UaxA7y5saIk , (accessed on 05.02.2020).

[M11] ***, Makino GmbH, D200Z 5-Axis Vertical Machining Center - product catalogue, M482Ed 1804/1 (V-T-D)

[M12] ***, Makino GmbH, EDAF2 sinker EDM, https://www.makino.eu/en-us/machine-technology/machines/sinker-edm/edaf2, (accessed on 01.04.2022).

[M13] Milodin N. L, Popa N. M., Tutoveanu M. and Artimon F. P. G., *Compression Testing of PA2200 Additive Manufactured Lattice Structures*, International Conference on Reliable Systems Engineering, pp 304-315, 2021, DOI: 10.1007/978-3-030-83368-8_30

[M14] ***, Materialise NV, Materialise Magics, Version 25.02

[N01] Nissim N., Dudaie M., Barnea I. and Shaked N.T., *Real-Time Stain-Free Classification of Cancer Cells and Blood Cells Using Interferometric Phase Microscopy and Machine Learning*, Cytometry, 2020, DOI: https://doi.org/10.1002/cyto.a.24227

[N02] Naboni R. and Kunic A., *Design and Additive Manufacturing of Lattice-based Cellular Solids at Building Scale*, XXI Congreso de la Sociedad Ibero-americana de Gráfica Digital, 2017, Chile, Vol. 3, Iss. 12, DOI: 10.5151/sigradi2017-058

[O01] Oros Daraban A. E., Negrea C. S., Artimon F. P. G. and al., A Deep Look at Metal Additive Manufacturing Recycling and Use Tools for Sustainability Performance, Sustainability, Vol. 11 (19): 5494, 2019, DOI: 10.3390/su11195494

[P01] Placko D., *Metrology in Industry: The Key for Quality*, French College of Metrology, translated into english by Barbier J., ISTE Ltd, London, 2006, ISBN-13: 978-1-905209-51-4.

[P02] Popovici C., Savii G. and Killman V., Tehnologia construcției de mașini, Editura Didactică și Pedagogică, Bucharest, 1967.

[P03] Paschotta R., article on 'Irradiance' in the Encyclopedia of Laser Physics and Technology, 1. edition October 2008, Wiley-VCH, ISBN 978-3-527-40828-3 (https://www.rp-photonics.com/irradiance.html accessed on 02.10.2021).

[P04] ***, Polytec Gmbh, *File:Michelson Interferometer scheme.png*, Werner Boeglin, Dept. of Physics, FIU, Miami, <u>https://en.wikipedia.org/wiki/File:Michelson Interferometer scheme.png</u> (accessed on 12.10.2021).

[P05] Pan F., Nie L., Bai Y., Wang X. and Wu X., *Geometric errors measurement for coordinate measuring machines*, IOP Conference Series: Earth and Environmental Science, 2017, Vol. 81, Iss. 1, DOI: 10.1088/1755-1315/81/1/012117

[R01] Rabinovich S., *Measurement Errors and Uncertainties Third edition*, Journal of Chemical Information and Modeling, 2005, Vol. 53, Iss. 9, ISSN 9788578110796.

[R02] Raghavendra N. and Krishnamurthy L., *Engineering metrology and measurements*, Oxford University Press, 2013, Vol. 1, Iss. 2, ISBN-13: 978-0-19-808549-2.

[R03] Radhakrishnan P., Subramanyan S. and Raju V., *CAD/CAM/CIM – Third edition*, New age international (P) Ltd., Publishers, New Delhi, 2008, ISBN (13): 978-81-224-2711-0.

[R04] Ramesh R., Mannan M. and Poo A.N., *Error compensation in machine tools- a review. Part II: Thermal errors*, International Journal of Machine Tools and Manufacture, 2000, Vol. 40, Iss. 9, pp. 1257-1284.

[R05] Repossi P., Learning Optics from the History, J. of Physics: Conference Series, 2019, Vol. 1221, doi: 10.1088/1742-6596/1221/1/012054.

[R06] Roychoudhuri C., Fundamentals of Photonics, 2008, SPIE Press, ISBN: 9780819471284, doi: 10.1117/3.784938.

[R07] Riehle F., Gill P., Arias F. and Robertsson L., *The CIPM list of recommended frequency standard values: Guidelines and procedures*, Metrologia, Vol. 55, Iss. 2, 2018, pp. 188-200, doi: 10.1088/1681-7575/aaa302.

[R08] Rayleigh, The interferometer, Nature, No. 1536, Vol. 59, pp. 533, 1899, doi: https://doi.org/10.1038/059533b0.

[R09] ***, Renishaw plc., *XC-80 compensator and sensors*, <u>https://www.renishaw.com/en/xc-80-compensator-and-sensors--8267</u>, (accessed on 14.10.2021).

[R10] ***, Renishaw plc., XL-80 system support disk, Version 2.1.2, 2008, United Kingdom.

[S01] Shah R. and Ward P., *Lean manufacturing: context, practice bundles, and performance*, Journal of Operations Management, Vol. 21 (2003), pp. 129-149.

[S02] Smith G. P., *Machine Tool Metrology – An Industrial Handbook*, Springer International Publishing Switzerland, 2016, ISBN 978-3-319-25109-7, DOI 10.1007/978-3-319-25109-7.

[S03] Schleicher M. and Blasinger F., *Control Engineering* - A guide for beginners, 3rd edition, JUMO GmbH & Co. KG, Germany, 2003, ISBN: 3-935742-01-0

[S04]Slocum A. R., Precision Machine Design, Society of Manufacturing, 1992, Vol. 02, Iss.2, DOI: 10.1016/0141-6359(80)90078-1

[S05] Steinmetz C. R., Sub-micron position measurement and control on precision machine tools with laser interferometry, Precision Engineering, 1990, Vol. 12, Iss. 1, pp. 12-24, DOI: 10.1016/0141-6359(90)90004-I

[S06] Shirinzadeh B., Teoh P. L., Tian Y., Dalvand M. M., Zhong Y. and Liaw H. C., *Laser interferometry-based guidance methodology for high precision positioning of mechanisms and robots*, Robotics and Computer-Integrated Manufacturing, Vol. 26, Iss. 1, 2010, pp. 74-82, DOI: https://doi.org/10.1016/j.rcim.2009.04.002

[S07] Sudatham W., Matsumoto H., Takahashi S. and Takamasu K., *Verification of the positioning accuracy of industrial coordinate measuring machine using optical-comb pulsed interferometer with a rough metal ball target*, Precision Engineering, Vol. 41, 2015, pp. 63-67, DOI: https://doi.org/10.1016/j.precisioneng.2015.01.007

[S08] Sarychikhina O., Gomez Palacios D., Delgado Argote L. A. and Gonzales Ortega A., *Application of satellite SAR interferometry* for the detection and monitoring of landslides along the Tijuana - Ensenada Scenic Highway, Baja California, Mexico, Journal of South American Earth Sciences, 2020, Vol. 107, DOI: https://doi.org/10.1016/j.jsames.2020.103030.

[S09] Shukla R. P., Udupa D. V., Das N. C. and Mantravadi M. V., *Non-destructive thickness measurement of dichromated gelatin films deposited on glass plates*, Optics and Laser Technology, Vol. 38, 2006, pp. 552-557 DOI: 10.1016/j.optlastec.2004.11.020

[S10] Steinberg I. Z. and Schachman H. K., Ultracentrifugation Studies with Absorption Optics. V. Analysis of Interacting Systems Involving Macromolecules and Small Molecules, Biochemistry, Vol. 5, 1966, pp. 3728-3747 DOI: 10.1021/bi00876a003

[S11] Schwenke H., Franke M. and Hannaford J., *Error mapping of CMMs and machine tools by a single tracking interferometer*, CIRP Annals - Manufacturing Technology, Vol. 54, Iss. 1, Pp. 475-478, DOI: 10.1016/S0007-8506(07)60148-6

[S12] Schede R. W, *Interferometers for Use as integral parts of machine tools*, IEEE Transactions on Industry and General Applications, Vol. 3, No. 4, 1967, pp. 328-332

[S13] Sommargren G. E., *Linear/angular displacement interferometer for wafer stage metrology*, SPIE 1989 Microlithography Conferences, Opticel/Laser Microlithography II, Vol. 1088, DOI: 10.1117/12.953154

[S14] **Stochioiu F. P. G.**, Ulerich O. and Rolea A., *Femoral stems with lattice structures integration: from concept to part*, 4th International Conference on Materials: Advanced and Emerging Materials

[S15] Stochioiu F. P. G., Stochioiu C., Rolea A. and Gannam N., *Interferometric evaluation of positioning error and position repeatability on CNCs*, IOP Conf. Ser.: Mater. Sci. Eng. 1268 012009, 2022, DOI 10.1088/1757-899X/1268/1/012009

[T01] Touloukian Y. S., Kirby R. K., Taylor R. K. and Desai P. D., *Thermal expansion; Metallic Elements and Alloys*, Thermophysical Properties of Matter, Vol. 12, Springer Science+Business Media, New York, 1975, ISBN 978-1-4757-1624-5.

[T02] ***, The Doppler Effect: a measurement technique fit for Nobel prizes, <u>https://rockyworlds.wordpress.com/2017/07/05/the-doppler-effect-a-measurement-technique-fit-for-nobel-prizes/</u> (accessed on 07.10.2021).

[T03] Tanaka M. and Nakayama K. A New Optical Interferometer for Absolute Measurement of Linear Displacement in the Subnanometer Range, Japanese Journal of Applied Physics, Vol. 22, No. 4A. DOI: 10.1143/JJAP.22.L233

[T04] Tratnik J. and Batagalj B., *Fiber interferometric communications using Mach-Zehnder interferometer topology*, Eurocon, 2009, doi: 10.1109/EURCON.2009.5167877

[T05] Tyler Estler W., *High-accuracy displacement interferometry in air*, Applied Optics, Vol. 24, No. 6, 1985, Pp. 808-815, doi: 10.1364/ao.24.000808

[T06] ***, Types of errors – 6. Other errors, <u>https://lasertex.eu/support/interferometer-usage-documentation/types-of-errors/</u> (accessed on 08.10.2021).

[T07] Tkaczyk T., Jozwicki R., *Heterodyne interferometer for measurement of linear displacements of a rough sample*, Tenth Polish-Czech-Slovak Optical Conference: Wave and Quantum Aspects of Contemporary Optics, Vol. 3320, Iss. 2, 1998, Pp. 147-152, doi: 10.1117/12.301329

[T08] Tang H., Duan J. and Zhao Q., A systematic approach on analyzing the relationship between straightness & angular errors and guideway surface in precise linear stage, International Journal of Machine Tools and Manufacture, Vol. 120, 2017, Pp. 12-19, doi: 10.1016/j.ijmachtools.2017.04.010

[T09] ***, Trinkle 3D GmbH and Schmalz GmbH – 3D Print Mass Customized Grippers Thanks to trinckle, <u>https://3dprint.com/259529/3d-print-mass-customized-grippers-thanks-to-trinckle/</u> (accessed on 10.02.2022).

[U01] Usamentiaga R., Venegas P., Guerediaga J., Vega L., Molleda J. and Bulnes F. G., *Infrared Thermography for Temperature Measurement and Non-Destructive Testing*, Sensors, 2014, Vol. 14, Iss. 7, pp. 12305-12348, ISSN: 1424-8220.

[V01] Vinod T. P., Metrology and Quality Control - For semester V, Third Year Diploma Courses in Mechanical and Production Engineering/Production Technology, Ed. 5, ISBN 978-93-83971-66-4.

[V02] Vlăsceanu D., Stochioiu C. and Gheorghiu H. M., *Noțiuni de mecanică și rezistența materialelor*, Bren, 2017, ISBN: 978-606-610-207-0.

[V03] Vala A, Mewada H., Patel A., Nagora U. and Pathak S., *Phase detection system based on digital signal processing in millimeter wave interferometer for fusion plasma diagnostics*, Fusion Engineering and Design, 2020, Vol. 157, DOI: https://doi.org/10.1016/j.fusengdes.2020.111637

[W01] Watts F. B., Engineering Documentation Control Handbook, Configuration Management And Product Lifecycle Management, Ed. 4, Elsevier, 2012, ISBN: 978-1-4557-7860-7.

[W02] White G. K. and Minges M. L., *Thermophysical Properties of Some Key Solids: An Update*, International Journal of Thermophysics, Vol. 18, No. 5, 1997.

[W03] Webster J. G. and Eren H., *Measurement, Instrumentation, and Sensors Handbook,* Spatial, Mechanical, Thermal, and Radiation Measurement, 2nd Edition, CRC Press, ISBN 978-1-4398-4889-0.

[W04] ***, "What is a Fizeau Interferometer?", https://4dtechnology.com/products/fizeau-interferometer/ (accessed on 12.10.2021).

[W05] Wirotrattanaphaphisan K., Buajarern J. and Butdee S., Uncertainty evaluation for absolute flatness measurement on horizontally aligned fizeau interferometer, Journal of Physics: Conference Series, Vol. 1183, Iss. 1, 2019, DOI: 10.1088/1742-6596/1183/1/012009

[W06] Wang D., Abdalla M. M., Wang Z. P. and Su Z., *Streamline stiffener path optimization (SSPO) for embedded stiffener layout design of non-uniform curved grid-stiffened composite (NCGC) structures*, Computer Methods in Applied Mechanics and Engineering - © 2018 Elsevier B.V., Vol. 344, pp. 1021 – 1050, 2019, DOI: 10.1016/j.cma.2018.09.013.

[X01] Xiang Yang. L. and Ettemeyer A., *Strain measurement by three-dimensional electronic speckle pattern interferometry: potentials, limitations and applications*, Optical Engineering, Vol. 42, Iss. 5, 2003, DOI: https://doi.org/10.1117/1.1566781

[X02] Xu J., Jahromi A. K., Brake J., Robinson J. E., and Yang C., *Interferometric speckle visibility spectroscopy (ISVS) for human cerebral blood flow monitoring*, APL Photonics, Vol. 5, 2020, DOI: https://doi.org/10.1063/5.0021988

[Y01] Yandayan T., Geckeler R., Just A., Siewert F., Grubert B., Prieto E., Perez M., Pisani M., Przybylska J. and Thalmann R., *Guidelines on the Calibration of Autocollimators*, Euramet Calibration Guide No 22, vers. 1.0, 07/2017.

[Y02] Yadavalli N. S., Advances in Experimental Methods to Probe Surface Relief Grating Formation Mechanism in Photosensitive Materials, University of Potsdam, Potsdam, April 2014.

[Y03] Yasin A. S. and Agus R., *Design of Fiber Optic Ring Resonator as a Gyroscope*, Physics Department, Faculty of Mathematics and Natural Sciences, Institute of Technology Sepuluh Nopember (ITS), 2013, <u>https://www.researchgate.net/publication/258697538_Design_of_Ring_Resonator_Fiber_Optics_as_a_Gyroscope</u> (accessed on 12.10.2021).

[Z01] Zihua Z., Chengwei L. and Zhen H., *Vibration measurement based on multiple Hilbert transform for self-mixing interferometry*, Optics Communications, Vol. 436, 2019, pp. 192-196, DOI: https://doi.org/10.1016/j.optcom.2018.12.032

[Z02] Zetie K. P., Adams S. F. and Tocknell R. M., *How does a Mach-Zehnder interferometer work?* Physics Education, Vol. 35, No. 1, 2000, DOI: https://doi.org/10.1088/0031-9120/35/1/308